Interactive effects of *Bacillus subtilis* and elevated temperature on germination, growth and grain quality of cowpea irrigated with acid mine drainage

by

THALUKANYO NEVHULAUDZI

STUDENT NUMBER: 58534601

A dissertation submitted in fulfilment of the academic requirements for the

degree of

MASTER OF SCIENCE IN ENVIRONMENTAL SCIENCES

in the

College of Agriculture and Environmental Sciences

Department of Environmental Sciences

at the

UNIVERSITY OF SOUTH AFRICA

Supervisor : Dr. Sheku Alfred Kanu

Co-Supervisor : Professor Khayalethu Ntushelo

February 2019

DECLARATION

I, Thalukanyo Nevhulaudzi, hereby declare that the work presented in this dissertation entitled "INTERACTIVE EFFECTS OF *BACILLUS SUBTILIS* AND ELEVATED TEMPERATURE ON GERMINATION, GROWTH AND GRAIN QUALITY OF COWPEA IRRIGATED WITH ACID MINE DRAINAGE" is original work done by myself under the mentorship of my supervisors. Additionally, I declare that the work presented herein has not been published or submitted at any other institution as part of the requirements for any degree programme. All cited literature in this dissertation from other individuals or institutions has been acknowledged and listed in the reference section. I also certify that I have complied with the rules, requirements, procedures and policies of the University of South Africa.

Candidate: THALUKANYO NEVHULAUDZI

Student Number: 58534601

Signature:

Date: 20 February 2019

ACKNOWLEDGEMENTS

I hereby wholeheartedly thank the almighty God for everything and all those who helped me during the course of writing this dissertation and completing my studies. The project was completed with assistance, advice, guidance and counselling from different people. Therefore, many thanks to my supervisors, Dr Sheku Alfred Kanu and Professor Khayalethu Ntushelo, for their determined provision of invaluable guidance and support towards this research. Without their skills and support, I wouldn't have completed the dissertation. Their advice and guidance from proposal writing through to fieldwork, laboratory work, and data analysis and ultimately, right through to the final write up of the dissertation are highly appreciated and loved.

Furthermore, acknowledgement is also due to my family as a whole, relatives and friends including colleagues at UNISA for their firm support and love, I cannot mention all their names as the list will be endless.

Finally, I wish to extend special thanks to all contributors towards completion of this project including the University of South Africa and National Research Foundation (NRF) for the provision of funds.

DEDICATION

To my loving family, father, mother, sister, brothers and son (Roana), I dedicate this dissertation.

TABLE OF CONTENT

DECLARATIONii
ACKNOWLEDGEMENTSiii
DEDICATIONiv
TABLE OF CONTENTv
LIST OF FIGURESxi
ABBREVIATIONS xii
ABSTRACTxiv
CHAPTER 1: GENERAL INTRODUCTION AND STUDY BACKGROUND1
1.1 INTRODUCTION1
1.2 BACKGROUND TO THE STUDY1
1.3 RESEARCH PROBLEM
1.4 AIM AND OBJECTIVES OF THE STUDY4
1.4.1 Research Aim
1.4.2 Research Objectives4
1.5 HYPOTHESES
1.6 SIGNIFICANCE OF THE STUDY
1.7 ETHICAL CONSIDERATION
1.8 STUDY LAYOUT
CHAPTER 2: LITERATURE REVIEW8
2.1 INTRODUCTION
2.2 ACID MINE DRAINAGE AS CONTAMINANT OF WATER SOURCES AND SOILS
2.3 MICROORGANISMS AS POTENTIAL BIO-REMEDIATORS OF ACID MINE DRAINAGE AND PLANT GROWTH ENHANCERS11
2.4 EFFECT OF TEMPERATURE ON GROWTH AND PHYSIOLOGY OF PLANTS.13
2.5 CLIMATE CHANGE AND CROP PRODUCTION 15

2.6 COWPEA CULTIVATION FOR FOOD SECURITY	18
2.7 CONCLUSIONS	19
CHAPTER 3: SEED GERMINATION INDICES OF COWPEA EXPOSED TO ELEVATED TEMPERATURE AND <i>BACILLUS SUBTILIS</i> (BD233) INOCULATIO	N20
3.1 ABSTRACT	20
3.2 INTRODUCTION	20
3.3 MATERIALS AND METHODS 3.3.1 Seed Materials and Measurements 3.3.2 Bacterial Strain	22 22 22
3.3.3 Germination Experiment and Experimental Design	22
3.4 RESULTS AND DISCUSSION	23
3.5 CONCLUSIONS	29
CHAPTER 4: GROWTH AND NUTRITIONAL RESPONSES OF COWPEA (Cv. Soronko) TO SHORT-TERM ELEVATED TEMPERATURE	30
4.1 ABSTRACT	30
4.2 INTRODUCTION	30
4.3 MATERIALS AND METHODS	32
4.3.1 Plant Material	32
4.3.2 Plant Growth Media and Experimental Conditions	32
 4.3.3 Fertilisation of Cowpea 4.3.4 Experimental Treatments: Biomass Determination and Nutrient Analysis 4.3.4.1 Cowpea growth and nutritional responses to elevated temperature	32 34 34 35
4.3.5 Determination of Nitrogen (Crude Protein) and Carbon4.3.5.1 Sample preparation4.3.5.2 Sample analysis	35 35 35
4.3.6 Data Analysis	36
4.4 RESULTS AND DISCUSSION	37
4.4.1 Cowpea Growth Response to Short-term Elevated Temperature	37

4.4.2 Effect of Short-term Elevated Temperature on Cowpea Crude Protein and Carbon Content
4.5 CONCLUSIONS41
CHAPTER 5: INTERACTION EFFECT OF <i>BACILLUS SUBTILIS</i> INOCULATION AND MINE WATER IRRIGATION ON COWPEA'S GROWTH, PHYSIOLOGY AND NUTIRIONAL QUALITY43
5.1 ABSTRACT
5.2 INTRODUCTION43
5.3 MATERIALS AND METHODS455.3.1 Plant Material455.3.2 Bacterial Strains45
5.3.3 Plant Growth Material and Cowpea Fertilization45
5.3.4 Co-Inoculation of Cowpea45
5.3.5 Acid Mine Drainage46
5.3.6 Experimental Treatments: <i>B. subtilis</i> Inoculation and AMD Irrigation46
5.3.7 Cowpea Growth, Nodulation and Yield Determination
5.3.7.2 Plant biomass (shoot and root weights)47
5.3.7.3 Nodulation and yield47
5.3.8 Measurement of Stomatal Conductance and Chlorophyll Content
5.3.9 Determination of Mineral Content in Mine Water (75% AMD), Cowpea Shoot and Grain48
5.3.11 Data Analysis48
5.4 RESULTS AND DISCUSSION
5.4.1 Interactive Effect of <i>Bacillus subtilis</i> (BD233) and Acid Mine Water Irrigation on the Growth, Nodulation and Physiology of Cowpea49
5.4.1 Effect of <i>Bacillus subtilis</i> (BD233) and Acid Mine Water Irrigation on the Nutritional Content of Cowpea's Shoot and Grain
5.5 CONCLUSIONS
CHAPTER 6: GENERAL DISCUSSION AND RECOMMENDATIONS
6.1 GENERAL DISCUSSION
6.2 RECOMMENDATIONS

EFERENCES71

LIST OF TABLES

Table (n=30)	3.	1:	Cow	pea	gen	otype	Se	ed	mea	surem	nents 24
Table 3. 2: <i>B. subtili</i> temperatur	Seed ge s (B+) es	ermina or	ation per without	centa <i>B.</i>	ge (G%) <i>subtilis</i>	of cow (B-)	pea ge and ii	notypes ncubate	s inocu d at	ulated diff	with erent 26
Table 3. 3: (B+) or temperatur	Seed ge without e	ermina B.	ation ind <i>subtilis</i>	ices o (B-)	f cowpea and ir	a genoty icubate	ypes in d for	oculate four	d with days	B. st at	ıbtilis 35°C 27
Table 3. 4: (B+) or temperatur	Seedlin witho es	g plur out	mule len <i>B.</i> ຣເ	gth of <i>ubtilis</i>	cowpea (B-)	genoty and	pes ind incu	oculated bated	d with at	B. st diffe	<i>ıbtilis</i> erent 29
Table 4. solutions	1: N-	free	plant	nutrie	ents an	d thei	ir con	centrat	ions	in s	stock 33
Table 4. 2: and dry we	Effects	of ter)) at c	nperatur lifferent (e on t growth	he plant stages	biomas of cowp	ss (g/pl ea gro\	ant) (fre wn in a g	esh we growth	eight chan	(FW) nber. 37
Table 5.1	The conc	entra	tion of he	eavy n	netals in	75% m	ine wa	ter			.49
Table 5.2: on the gro	Effects c owth par	of <i>B.</i> s ramet	<i>ubtilis</i> in ers of c	oculat cowpe	tion and a genot	acid mi ypes at	ne wat t 125	er (75% DAP in	a gl) irrig asshc	ation ouse. 52
Table 5.3: irrigation	Interactiv on the ç	ve effe growtł	ects of <i>B</i> param	<i>. subti</i> ieters	lis inocu of cow	ation ar pea at	nd acid 125 I	mine w DAP in	ater (7 a gl	75% A asshc	AMD) buse. 53
Table 5.4: on nodula	Effects c tion and	of <i>B.</i> s d yiel	<i>ubtilis</i> in d of co	oculat wpea	tion and genoty	acid mi pes at	ne wat 125 [er (75% DAP in	a gl) irrig assho	ation ouse. 55
Table 5.5: irrigation o	Interactiv n nodula	ve effe	ects of <i>B</i> and yield	. subti of co	<i>lis</i> inocu wpea ge	ation ar notypes	nd acid s at 12	mine w 5 DAP i	ater (7 in a gl	75% A assho	AMD) buse. 56
Table 5.6: irrigation o at 77 DAP	Interacti n leaf sto	ve ef omata	fects of I conduc	<i>B. su</i> tance	<i>btilis</i> ino (m²/s m	culation ol ⁻¹) of (n and r cowpea	nine wa a grown	ater (7 in a g	5% A lassh	AMD) ouse 58
Table 5.7: irrigation c	Interactiv on leaf c	/e effe hloro	ects of <i>B</i> phyll co	. subti ntent	<i>lis</i> inocu (mg/m²)	ation ar of cov	nd acid vpea g	mine w rown ir	ater (7 n a gl	75% A assho	AMD) buse. 59
Table 5.8: matter (Effect of DM))	inocu of	Ilation ar cowpea	nd irrig ge	ation or notypes	the mir unde	neral co er gl	omposit asshou	ion in se	shoot condi	t (dry tions 61

Table	5.9:	Effect	of ir	noculation	and irr	igation of	on the	mineral	composition	in grain (dry
matter	· (I	DM))	of	cowpe	a ge	enotype	s un	der g	glasshouse	conditions
										62

LIST OF FIGURES

Figure 4.4: Effect of elevated temperature on the nutritional content (Carbon value (%)) of cowpea cv Soronko shoot at different growth stages.......40

Figure 4.5: Effect of elevated temperature on the nutritional content (C/N ratio) of cowpea cv Soronko shoot at different growth stages......40

ABBREVIATIONS

ANOVA	 Analysis of Variance					
AMD	 Acid Mine Drainage					
B.subtilis	 Bacillus subtilis					
Ca	 Calcium					
Cd	 Cadmium					
cm	 Centimetre					
Cr	 Chromium					
Cu	 Copper					
DAP	 Days after Planting					
DAS	 Days after Sowing					
DWAF	 Department of Water Affairs and					
	Forestry					
FAO	 Food and Agriculture Organisation					
Fe	 Iron					
G%	 Germination Percentage					
GCMs	 General Circulation Models					
GHG	 Greenhouse Gas					
GI	 Germination Index					
GRI	 Germination Rate Index					
h	 Hour					
Hg	 Mercury					
IPCC	 Intergovernmental Panel on					
	Climate Change					
К	 Potassium					
Mg	 Magnesium					
Min	 Minute					
mL	 Millilitre					
mm	 Millimetre					
mМ	 Millimolar					
Mn	 Manganese					
Ni	 Nickel					
°C	 Degree Celsius					

Pb	 Lead			
PGPR	 Plant Growth Promoting Bacteria			
UNFCCC	 United	Nations	Framework	
	Conventio	on on Clima	te Change	
UNISA	 University of South Africa			
Zn	 Zinc			

ABSTRACT

This study's main goal was to evaluate Bacillus subtilis inoculation and mine water irrigation effect on germination, growth, nodulation, physiology and shoot/grain guality of cowpea genotypes exposed to extreme climatic conditions (elevated temperatures). The first experiment evaluated the interactive effect of Bacillus subtilis (BD233) inoculation and elevated temperature on germination indices and plumule lengths of three genotypes (Asetanapa, Soronko and Nyira) of cowpea. The results showed that interaction between *B. subtilis* (BD233) and temperature significantly (p<0.05) influenced the germination indices (germination percentage (G%), germination index (GI) and germination rate index (GRI)) and plumule length of cowpea seedlings and genotype responses were significantly different. At elevated temperature (35°C), inoculation with *B. subtilis* (BD233) enhanced seed germination and growth of cowpea. The second experiment evaluated the effect of temperature on growth and nutritional content of cowpea incubated for seven days in a growth chamber. The results showed that when cowpea genotype, Soronko, was incubated at different temperature regimes, the whole plant biomass, shoot carbon and crude protein contents were significantly affected with temperature increases at all three stages of the plants' life cycle. The results suggest that the pre-flowering (40 DAP) and flowering (90 DAP) stages of cowpea compared to post-flowering (123 DAP) are more susceptible to elevated temperatures (30-35°C). The third experiment evaluated Bacillus subtilis inoculation and mine water irrigation effect on growth, nodulation, physiology and nutritional content of cowpea under glasshouse conditions. The results revealed that the interaction of *B. subtilis* (BD233) inoculation and mine water (75% AMD) irrigation was significant for the growth, nodulation, stomatal conductance, chlorophyll contents and shoot/grain nutritional quality of cowpea genotypes. In comparison with control, generally, B. subtilis inoculation enhanced the growth, nodulation and yield of all tested cowpea genotypes and irrigation with mine water significantly influenced the mineral contents in both shoot and grain of cowpea. Taken together, findings in this study have implications for cultivation of cowpea, an important candidate for food/nutrition security in Africa, under future climate change scenarios.

CHAPTER 1: GENERAL INTRODUCTION AND STUDY BACKGROUND

1.1 INTRODUCTION

Water for irrigation of crops is a scarce resource and most surface waters are being contaminated due to mining activities, especially in developing countries such as South Africa, a water-scarce country with a long mining history. The advent of climate change has compounded the problem with frequent occurrences of droughts, erratic rainfall patterns (less rains) and elevated temperatures, which have the potential to impact the growth of most crops. This potential threat to the security of household food and nutrition, posed by climate change and elevated temperatures, is worrisome and therefore, there is a need to clean up contaminated surface waters especially acid mine water and mitigate climate change. Microorganisms including some *Bacillus* spp. have been reported as bio-remediators of contaminated waters and soils (with heavy metals) and also as plant growth enhancers.

The overall goal of this study was to evaluate the interactive effect of *Bacillus subtilis* and mine water irrigation on the germination indices, nodulation, growth, physiology and nutritional quality of three cowpea genotypes exposed to elevated temperature regimes. This chapter provides the study background, research problem, aim and objectives, significance of the study, research hypothesis, and the overall organization of the dissertation.

1.2 BACKGROUND TO THE STUDY

Acid mine drainage (AMD) generated from mining activities remains to be a significant water pollution problem worldwide (Coetzee et al., 2010; Naidoo, 2017). Exposure to air and water of sulphide bearing minerals, often in the form of pyrite, generates AMD. The oxidation of sulphide in AMD produces acids such as H₂SO₄ and heavy metals (Fe, Zn, Mn, Pb, Cd, Cu, Ni). AMD generated through entrance of water into mine voids, is characterized by the following: low pH (2-4), high sulphur concentration (1-2 g/l) and levels of heavy metals (Fe, Zn, Mn, Pb, Cd, Cu, Ni, Hg and Cr) (Vadapali et al., 2008). These heavy metals in water systems can persist and bio-accumulate in living organisms such as plants, especially if plants are exposed to higher concentrations over time and thereafter consumed by animals and humans in the food web, ultimately posing health hazards and environmental risk in the ecosystem

(Naicker et al., 2003). The discharge of heavy metals into aquatic ecosystems thereby causing contamination of surface water and groundwater systems has become a matter of concern (Mamba et al., 2009). In addition, changes in pH brought about by AMD tends to significantly impact the chemistry of mine waters and rivers into which they are discharged. Based on the metal being mined and mining process, the pH level can vary from one area to another, for instance, a comparison of the pH of surfaces water from the precincts a gold (extremely acidic (pH<2)) and coal (pH >4)) mines showed significant differences (Nevhulaudzi et al., 2014).

In South Africa, mining activities have a long history and has left behind huge amounts of AMD in the mining industry's precincts. The mining activities associated with income generation opportunities have attracted a huge population of migrant workers in these mining areas. This development has led to increased demand for water from surface waters by the resident population for both industrial and domestic uses including irrigation of crops. In South Africa, the close proximity of mining industries to water catchment areas and therefore surface waters is an additional threat to ecosystems due to pollution by AMD derived from the pyritic materialisations in coal and gold deposits. The country has a little arable land percentage and the rainfall pattern is erratic and low annually, therefore the country is considered water-scarce (Jovanovich et al., 1998). In addition, most surface waters are polluted by wastewaters from industries and urbanization including municipal waste (Du Plessis, 1983). Therefore, good quality water scarcity in the country especially for crop irrigation by farmers is of major concern. Recently, the country experienced droughts in some provinces (Western Cape and Free State) and the looming climate change due to the El Nino phenomenon makes the situation very worrisome. One of the main threats of climate change is rise in global average temperatures, which poses additional problems for seed germination, crop development and other living organisms (Ewing, 1981). Furthermore, erratic rain patterns (less rains) and frequent occurrence of droughts are also associated with climate change. Therefore, the possible utilization of available AMD contaminated surface waters for agricultural crop irrigation and the impending physiological threats due to increases in temperature/climate change are subjects of interest to the scientific community in South Africa in particular and the world in general. In South Africa, the amelioration of acid mine water using lime for the irrigation of crops has been reported (Du Plessis, 1983). However, the amelioration of AMD water with lime is not only expensive but can also release pollutants in soils following irrigation of crops with lime-treated acid mine water. On the other hand, although

reverse osmosis has been demonstrated to possibly treat badly polluted water such as acid mine water to drinking quality and crop irrigation standards, the cost of ameliorating mine water by this process can be however colossal (McCarthy, 2011). Thus, there is a dire need to search for cheaper ameliorating alternatives or available bio-remediators to de-contaminate AMD water without posing any threat to ecosystems or agronomic systems. At present, clean-up processes of heavy metal pollution are costly and environmentally damaging. Interestingly, researchers have started generating cost-effective technologies that include the use of microorganisms for the cleaning process of pollutants in mine water (Abou-Shanab et al., 2003; Hooda, 2007), for possible use in crop irrigation. One good example of such technologies is bioremediation. Bioremediation is a technology that uses organisms to breakdown contaminants and thereby decontaminate polluted natural environments (Agarwal, 1998).

Cowpea (*Vigna unguiculata* L. Walp) is an important nutritious crop that belongs to the family Fabaceae /Leguminosae. It is reported to have evolved in West Africa (Smartt, 1985) but nowadays it is cultivated in most regions of the world including East and Southern Africa, Southeast Asia, Southern United States and Latin America. The crop is mostly cultivated because of its ability to contribute nitrogen to nutrient poor soils in addition to providing cheap dietary protein, carbohydrate, minerals and vitamins for humans (Bressani, 1985, Belane, 2011). Cowpea grain is rich in protein (42%) and carbohydrate (57%) and the green leaves and young pods are capable of suppling up to 35% protein when consumed as vegetables (Kadam et al., 1984; Belane, 2011). In South Africa, the crop is cultivated and consumed mostly by smallholder farmers in rural communities in all provinces in the country and therefore, has a potential role to play in securing household food and nutrition.

1.3 RESEARCH PROBLEM

South Africa is a water-scarce country and climate change will compound the water crisis as evident by the recent droughts in some regions in the country (Western Cape, Free State and Mpumalanga provinces). However, the country has huge deposits of minerals with pyritic formations, which is the source of AMD pollution in most surface waters including dams and rivers due to its long mining history especially in the

Witwatersrand Basin (McCarthy, 2011). AMD water is toxic (with heavy metals and low acidity) and when used for irrigation can affect the growth and nutritional quality of crops. As well, the exposure to extreme climatic conditions (such as elevated CO₂ and temperatures) poses a threat to the growth and yield of crops such as cowpea and vegetables. Increases in global temperatures (elevated temperature) caused by climate change can influence seed germination and growth of important economic crops such as cowpea thereby posing threats to future household food and nutrition security. In future, there would be changes in agricultural crop production/food security due to climate change as a results of high temperatures, which can have adverse effects on crop yields because of the overall reduction in seed germination and plant growth (Ewing, 1981). Furthermore, the need to mitigate climate change cannot be over emphasized especially in developing countries in sub-Sahara Africa such as South Africa, which is a water-scarce country. There is a dire need to ameliorate AMD water with cheaper alternatives that are environmentally-friendly and readily available. The use of microorganisms in the amelioration of badly polluted water such as AMD has been proposed. Therefore, the suitability of *B. subtilis* as a potential heavy metal resistant bacteria in soils polluted with mine water was explored in this study.

1.4 AIM AND OBJECTIVES OF THE STUDY

1.4.1 Research Aim

The main goal of this study was to evaluate seed germination, growth, nodulation, physiology and shoot and grain quality of cowpea genotypes inoculated with *Bacillus subtilis* and irrigated with mine water and exposed to extreme climatic conditions (elevated temperatures).

1.4.2 Research Objectives

- To assess and compare the seed germination indices and growth of three cowpea genotypes (Asetenapa, Soronko & Nyira) inoculated with *B. subtilis* (BD233) and exposed to elevated temperature regimes.
- To determine the growth and shoot/grain nutritional quality responses of cowpea to elevated temperatures (>30°C).

- To evaluate the interactive effect of inoculation with *B. subtilis* (BD233) and mine water irrigation on the growth, nodulation and physiology (stomatal conductance and leaf chlorophyll content) of cowpea genotypes.
- To assess and compare the shoot and grain nutritional quality of the three cowpea genotypes irrigated with mine water and inoculated with *B. subtilis* (BD233).

1.5 HYPOTHESES

- Inoculation with *B. subtilis* (BD233) and exposure to elevated temperatures have no effect on the seed germination indices and growth of three cowpea genotypes.
- There are no changes in the growth and shoot/grain nutritional quality responses of cowpea to elevated temperatures.
- Interaction of *B. subtilis* (BD233) inoculation and mine water irrigation has no effect on the growth, nodulation and physiology of cowpea genotypes.
- There are no differences in the nutritional content of shoot and grain of cowpea genotypes irrigated with mine water and inoculated with *B. subtilis* (BD233).

1.6 SIGNIFICANCE OF THE STUDY

The practice of utilizing *Bacillus* species in agriculture is on the increase steadily mainly because it offers a substitute to expensive environmentally-unfriendly fertilizers (Bhattacharyya & Jha, 2012). Certain strains of *Bacillus* spp. are described to stimulate plant growth mainly by secreting siderophores and phytohormones e.g. auxins, cytokinins, gibberellins etc.) in plant rhizosphere soils (Ali et al., 2009a, Galaviz et al., 2018) or augmenting nutrient utilization (solubilisation of minerals) in plants (Bowen & Rovira, 1999; Zhang et al., 2010; Meng et al., 2016). Studies have found that *B. subtilis* can quench soil ills (such as nutrient deficiencies) and activities of plant pathogens to boost plant growth (Kloepper et al., 2004; Adesemoye et al., 2008; Felici et al., 2008). On the other hand, elevated temperatures due to future climate change can drastically affect the seed germination and growth of crops (Ewing, 1981) such as cowpeas. However, few or no studies have explored the interactive effect of *B. subtilis* and elevated temperatures on seed germination and growth of cowpea genotypes, an important crop in Africa and other regions in the world (Freire-Filho, 2011). Findings from this study have implications for the choice of cowpea genotypes grown especially

by resource-poor farmers in areas experiencing climate change (elevated temperatures) and irrigated with ameliorated AMD water. Among the tested cowpea genotypes, Soronko was revealed to have superior traits with regards to adaptation to climate change. At elevated temperatures (30-35°C) and without *B. subtilis* inoculation, Nyira followed by Soronko had the highest germination percentage, fastest germination rates (lowest GI & GRI values) and enhanced growth. Co-inoculation of Soronko with *B. subtilis* and a commercial inoculant, enhanced nodulation and yield compared to control even when irrigated with mine water. The interaction of *B. subtilis* inoculation and mine water irrigation (75% AMD) had a significant effect on the nutritional quality of cowpea. This study has demonstrated the potential of *B. subtilis* as a suitable metal resistant bacteria in soils polluted with acid mine water.

1.7 ETHICAL CONSIDERATION

This study was carried out in line with UNISA's policy on ethics, the research proposal was approved by CAES committee on 04 November 2016 (2016/CAES/093) and the approval was reviewed annually until the completion of the experiment.

1.8 STUDY LAYOUT

This dissertation consists of six chapters including three research chapters. Each chapter summary is presented here to guide the readers through the deliberations on topics presented.

Chapter one provides a background to the study and presents the aim and objectives for undertaking the study, including the research problem and significance of the study.

Chapter two deliberates the literature studied in relation to crop production and climate change and the role of microorganisms in bioremediation of AMD polluted water sources. It offers the purpose of conducting a literature review and findings from the literature in relation to the topic.

Chapters three, four and five present research findings for separate experiments. Specifically, chapter three describes and discusses findings on the effect of elevated temperature and *Bacillus subtilis* inoculation on cowpea seed germination indices and growth. Chapter four describes and discusses findings on the effect of elevated temperature (>30°C) on the growth and nutritional quality of cowpea at three growth

stages. Chapter five describes and discusses findings on the interactive effect of *Bacillus subtilis* inoculation and mine water irrigation on the growth, nodulation, physiology and nutritional content of shoot and grain of cowpea genotypes under glasshouse conditions.

Chapter six presents the general discussion and recommendations thereof.

CHAPTER 2: LITERATURE REVIEW

2.1 INTRODUCTION

The previous chapter gave the background and research problem as an overview of the study. It also highlighted the study aim and objectives including its significance and concludes with the structure of the dissertation. This chapter is a review of the literature related to irrigation of crops with contaminated water and role of microorganisms in bioremediation of the badly polluted water such as AMD water. The literature on climate change (elevated temperature) effects on plant physiology and crop production and cowpea cultivation as food/nutrition security are also reviewed. The chapter closes with a summary of the gaps identified in the literature reviewed.

2.2 ACID MINE DRAINAGE AS CONTAMINANT OF WATER SOURCES AND SOILS

Surface waters are being contaminated due to mining activities resulting in clean water for crop irrigation becoming a very scarce commodity. Mining of gold and coal is associated with AMD responsible for long-term loss to watercourses and biodiversity including entire ecosystems (Akcil & koldas, 2006). Mostly, AMD occurs as a result of reactive sulphide minerals (pyrite) stored in waste rock dumps reacting with water and oxygen (Streren-Joyce et al., 2013). Acid mine drainage has high specific conductivity, concentrations of iron, aluminium, and manganese, and low pH (2-4), and concentrations of toxic heavy metals (Pb, Hg, Cr, Zn & Ni) (Streren-Joyce et al., 2013).

In developing countries such as South Africa, wastewater disposal has become problematic because increased urbanization and industrialization are producing large quantities of municipal waste and industrial effluent (acid mine drainage) (Alloway & Ayres, 1993). In South Africa, the disposal of AMD derived from mining activities is a serious environmental problem. Among the problems of disposing mine water from mines (gold and coal) and old underground workings, is the acidification of water sources and soils when mine water is discharged from the mine treatment plants into the environment over time. Acid mine drainage water is highly acidic (pH ranging typically from 2 to 4.4) and can therefore increase soil acidity. According to statistics provided by Venter et al. (2001), a huge proportion (about 5 million ha) of soils in South Africa are severely acidic and about 11 million ha are moderately acidic. Considering the country's long gold and coal mining history and vast deposit of pyritic formations,

it is highly probable that the prevalence of the reported high soil acidity may be due to AMD reaching water pathways or river catchments. Soil acidification due to either AMD or natural driving forces (hydrolytic displacement of base cations or oxidation reactions) can change soil properties and processes such as contraction of cation exchange capacity and expansion of anion exchange capacity (Fey, 2001). In addition, it can decrease the concentrations of some important basic cations (Ca, Mg, K) and solubility of nutrients (phosphorous and molybdenum) essential for the growth of plants and can also increase the concentration of some toxic elements (aluminium and manganese) (Sumner et al., 1990). The loss of essential nutrients in soils exposed to AMD and soil accumulation of heavy metals will translate into poor productivity if such soils are used for crop cultivation.

In most countries, one of the possibilities for disposing wastewater/mine water is to irrigate crops in agricultural land (Oloya & Tagwira, 1996). Agriculture utilises about 70% of water withdrawal hence it is expected that in times of water scarcity farmers will turn into wastewater/mine water as water source especially by poor-resource farmers in rural communities (United Nation's Food and Agriculture Organization (FAO), 2010). Long-term utilization of wastewater for irrigation of important economic crops such as cowpea may result in heavy metal accumulation in humans through the food web and contamination of soils in particular and the environment in general (Chipo et al., 2011). Long-term irrigation of agricultural soils with wastewater/mine water will ultimately results to soil contamination in addition to compromising the safety and quality of food crops (Muchuweti et al., 2006). Soil health may be endangered when wastewater/mine water are used for crop irrigation (Barman et al., 2000). Dougherty and Hall (1995) also stated that the use of wastewater/mine water in agricultural lands for a long-term has resulted to soil contamination with metals. Clearly, irrigation of crops with wastewater/mine water can bring changes in the guality of soil as trace element contributions are sustained for a long period. Heavy metals in wastewater/mine water are one of the factors negatively affecting natural environments (Ahmad Wani et al., 2007; Marchel et al., 2015). Heavy metal contamination in agricultural soils has become a thoughtful risk to global sustainability of crop production (Chen et. al., 2007; Ahmad Wani et al., 2007). These heavy metals include Cd, Pb, Zn, Cu, Hg, Cr, Mn, Fe, Zn & Ni.

Amongst the heavy metals, some are non-biodegradable and can persist in the environment over a long period. For example, cadmium is of a great concern due to

its toxic effects on flora and fauna even at low concentrations (Hu et al., 2009). Cadmium is categorised by high toxicity and high capacity to bio-accumulate in plants because it is persistent and non-biodegradable. Cadmium has a high phytotoxic impact when accumulated even at low concentrations and the impact on plants ranges from a reduction in growth, increased chlorosis and wilting to death of cells (Gallego et al., 2012). In addition, this heavy metal has a comparative high movement within the soil-plant system, which suggests it can go through the food chain and subsequently, build-up to toxic levels in living organisms thereby posing threats to their health in particular and that of the ecosystem in general (Burger, 2008) because it is persistent over a long period. A previous study found that high concentrations of cadmium in soils can arise naturally however often are due to industrialization or over use of fertilizers containing cadmium (Tiller et al., 1997). Lead is another heavy metal of great concern as it can also bio-accumulate and thereby cause health-related problems in living organisms including humans over time when consumed through the food chain. The uptake of heavy metals from the rhizosphere by plants is the main path of exposing humans to them in soils (Wang et al., 2007). However, irrigation of crops with ameliorated mine water or wastewater has its benefits, which include but not limited to water conservation. In addition, there are beneficial heavy metals e.g., Zn, Mn & Fe for plant growth especially when at the adequate concentrations but others (e.g. Cd, Pb) are not, even at low concentrations, because after their addition into soils they could be transferred through the food chain to humans and over time can bio-accumulate and persist and cause harmful effects when they reach toxic levels (Aziz et al., 1996). Recognition of the detrimental effects of some heavy metals has prompted the need to clean-up or ameliorate acid mine water or wastewaters containing heavy metals. Environmentally friendly processes need to be established and implemented to clean-up and protect the environment by means of bioremediation techniques (Stratten, 1987). One such environmentally-friendly process is the use of microorganisms in the amelioration of badly polluted waters such as AMD. Interestingly, some microorganisms are also known to enhance plant growth (Compant et al., 2010; Dary et al., 2010; Román-Ponce et al., 2017) through several mechanisms.

2.3 MICROORGANISMS AS POTENTIAL BIO-REMEDIATORS OF ACID MINE DRAINAGE AND PLANT GROWTH ENHANCERS

Bioremediation is a recent technology gaining popularity in the clean-up of polluted natural environments. The technology is considered as one of the most cost-effective and environmentally conscious management tool to recover polluted soils in their natural stance and manage the polluted environment. The technology is relatively cheaper than conventional methods (reverse osmosis, chemical fixation, physical treatment) because it relies on promoting the growth of indigenous microorganisms capable of breaking down contaminants on-site (Agarwal, 1998). Bioremediation process involves transmuting (biotransformation and biodegradation) contaminants into non-hazardous or less hazardous products (Singh et al., 2014). Biotransformation involves any modification of the structure of molecules/compounds by microorganisms into a less harmful molecule/compound. Biodegradation process involves the breaking down of a molecule/compound into environmentally-friendly or non-toxic products. Heavy metal resistant bacteria having plant growth promotion traits are currently being used as eco-friendly and less cost-effective technologies for bioremediation of metal contaminated soils (Khan et al., 2009). Microorganisms capable of biodegrading or bioremediation are recognized members of microbial consortiums and they include Bacillins (Singh et al., 2014), rhizobacterial strains of Alcaligenes, Bacillus, Curtobacterium, and Microbacterium, with plant growth-enhancing traits (Román-Ponce et al., 2017). In particular, legume plants associated with microorganisms (plant growth-promoting rhizobacteria (PGPR)) have been used for bioremediation of soils contaminated with heavy metals (Carrasco et al., 2005; Zhuang et al., 2007). For example, Bacillus PZ3 is a PGPR, was reported to reduce Cr in pea plants when inoculated in chromium amended soil and also enhance the plant's growth, nodulation, leghaemoglobin, chlorophyll and protein content compared to control plants (Wani & Zainab, 2016). These bacteria (PGPR) types are suggested to have the ability to secrete plant stimulating metabolites in heavy metal polluted soils (Wani et al., 2007) and in the process accelerate phytoremediation of such soils by enhancing the health and growth of plants (Compant et al., 2010; Dary et al., 2010). In essence, they can be referred to as metal resistant bacteria because they have the ability to effectively sequester metals in heavy metal contaminated soils and also enhance plant growth (Fatnassi et al., 2014).

Bacillus subtilis is a ubiquitous Gram-positive, rod-shaped bacteria noted to enhance the development and growth of plants including providing defense against pathogens. Microorganisms with the ability to grow in the rhizosphere soils of plants are best for use as biocontrol agents since the rhizosphere serves as an interface for pathogens to attack plant roots. Many bacteria species, including Bacillus, are thought to have the potential for biocontrol against plant pathogens (Weller, 1988). There are reports of Bacillus spp. having the ability to control diseases on a wide variety of plant species and also bio-remediate contaminated soils (Sheng & Xia, 2006). Furthermore, the biological control approach using microorganisms was successfully used to control many potato diseases and the possibilities for biological control of the potato brown rot disease have been studied by many workers (Weller, 1988; Jacobsen, 2002; Verma et al., 2010). Many rhizosphere organisms referred to as "plant growth promoting rhizobacteria (PGPR)" are noted to positively influence plant growth and plant health (Schippers, 1992). Studies in the Netherlands suggest that PGPR stimulated potato growth mainly by quashing cyanide-producing compounds (Weller, 1988). There are few benefits for applying PGPR mechanism thus include but not limited to increases in germination rates, nodulation, etc. An earlier study showed that chickpea seed inoculation with PGPR increased nodulation, N uptake, growth and yield (Verma et al., 2010). PGPR can also be used for biocontrol of plant disease and phytoremediation of soils polluted with heavy metals. Inoculation with PGPR and rhizobium has been shown to increase the biomass of shoot and root, N2 fixation and grain yield in chickpea (Verma et al., 2010). For example, inoculation of lentil with the Rhizobacterium strain RL9 in Zn contaminated soils caused an increase in dry matter (150%), nodule numbers (15%), nodule dry mass (27%), leghaemogloblin (30%), seed vield (10%) and grain protein (8%), compared with control (Ahmad Wani et al., 2007). However, climatic variability has been reported to impact the effectiveness of PGPR (Okon & Labandera-Gonzalez, 1994) especially under unfavourable field conditions, where they are expected to function normally such as in agricultural systems. Studies under different climatic conditions conducted by Bustamante and Ciampi (1989) also described Pseudomonas fluorescens as a soil treatment that protected 90% of potato seedlings against infection with bacterial wilt disease and lowered the population of the pathogenic bacteria. Other studies have demonstrated the potential of Pseudomonas strains to control soil-borne pathogens and enhance growth and yield of potato and radish plants (Burr et al., 1978; Klopper et al., 1980; Howie & Echandi, 1983). In addition, in a study by Prathibha and Siddalingeshwara (2013), they reported

that *Pseudomonas fluorescens* and *Bacillus subtilis* considerably improved seed germination, vigor index and nutritional quality (protein and carbohydrate content) of sorghum. In a recent study, Román-Ponce et al. (2017) reported that inoculation of *Brassica nigra* seeds with microorganisms such as strains of *Microbacterium* sp., and *Curtobacterium* sp. considerably improved seed germination and root development in soils loaded with heavy metal (Zn). These authors concluded that these rhizobacterial strains have potential as efficient bio-inoculants in phytoremediation of multiple heavy metals contaminated soils (Román-Ponce et al., 2017). However, the suitability of *B. subtilis* as a metal resistant bacteria especially in soils polluted with AMD water is yet to be explored including the impact of temperature on its activities as variations in climatic conditions can affect the activities of microorganisms and a plant's physiology as well (Okon & Labandera-Gonzalez, 1994).

2.4 EFFECT OF TEMPERATURE ON GROWTH AND PHYSIOLOGY OF PLANTS

By 2100, the current global temperature is predicted to increase by 1.8-4°C (Intergovernmental Panel on Climate Change (IPCC), 2007). Among environmental factors, temperature is considered as the most detrimental stress affecting plant growth and development. This is the case because numerous biochemical processes involved in plant activities are sensitive to heat stress due to elevated temperature. However, when exposed to changes in temperature regimes, plants have coping mechanisms that involve stress-induced biochemical and physiological alterations including shifts in primary and secondary metabolites profiles (Kaplan et al., 2004). The responses and coping mechanisms of plants to temperature vary across genera, species and sometimes between genotypes of the same species.

Generally, temperature is noted to affect various processes involved in a plant's growth and development starting from germination to seedling emergence and establishment and through to plant maturity including the yield and other biochemical and physiological processes. Therefore, it is important to investigate the effect of temperature on physiological and biochemical processes including photosynthesis, chlorophyll content, and stomatal conductance in order to enhance the productivity of crops exposed to elevated temperature (>30°C). For instance, pollination is severely affected in maize with exposure to an elevated (35°C) temperature from 30°C. Pollen viability was noted to decrease significantly with maize exposure to a temperature above 35°C (Dupuis & Dumas, 1990) and exposure to temperatures above 30°C was

noted to decrease grain yield in maize (Commuri & Jones, 2001) and rice (Kim et al., 2013).

So far, several investigations have established that amongst the environmental factors, temperature significantly stimulated seed germination (Rivers & Webber, 1971; White & Monstes, 1993; Craufurd et al., 1996), and the rate of seed germination and seedling emergence is influenced substantially by temperature (Garcia-Huidobro et al., 1982; Covell et al., 1986). For example, low temperatures at higher latitudes and highland tropical sites frequently caused a low percentage of seed germination resulting in poor stand formations (White & Monstes, 1993). Other studies showed that at 5°C or 10°C cowpea seeds failed to germinate however, at higher temperature (20°C-30°C), maximum germination (90%) was reached and only 85 to 88% at these temperatures; 15°C, 35°C and 40°C (Covell et al., 1986). Similar results were obtained showing that cooler temperatures (<15°C) reduced cowpea seeds germination rate and above 80% seed germination was obtained at constant temperatures (18°C-40°C), whereas elevated temperatures (>40°C) severely reduced seed germination and above 43.5°C, there were zero percentage of seed germinations for all genotypes (Craufurd et al., 1996). It is important to highlight that seed storage environment greatly influences the period of seed survival and the optimal temperature for germination is significantly affected by previous treatment of seed or its age (Auld & O'Connell, 1991). A study conducted on the effect of temperature seed germination of common bean found that among 16 genotypes, the base temperature varied from 5-8°C to 8-9°C with lowest germination percentage whereas optimum temperature ranged from 25-5°C to 29-3°C and provided high germination percentages (White & Monstes, 1993). It was also revealed that temperature affected seed germination rate: at 4.4°C germination was delayed until the 26th day but at 26.6°C, the germination began within one day (Rivers & Webber, 1971). Further reports showed that high temperatures generated by a fire could promote germination in species with a hard seed cover, by inducing the breaking of seed coats, thereby facilitating the subsequent embryo imbibition and radicle expansion (Auld & O'Connell, 1991).

In other studies, elevated temperatures have been noted to influence potato plant's growth and yield (Hijmans, 2003; Levy & Veilleux, 2007; Hancock et al., 2014) because several processes in potato physiology are impacted. For instance, the processes affected include tuber development, carbon transport/partitioning, CO₂

fixation, chlorophyll content and the photosynthetic performance of the potato plant in general thus leading to lower yields (Hancock et al., 2014). Furthermore, in response to elevated temperature there can be shifts in the leaf and tuber primary and secondary metabolites profiles of potato (Hancock et al., 2014). It is important to note that elevated temperature is associated with climate change, which if not mitigated, will significantly impact household food and nutrition security especially in poor rural communities in developing countries such as South Africa and other African countries.

2.5 CLIMATE CHANGE AND CROP PRODUCTION

In general, climate change may denote fluctuations in average weather conditions, or weather variations within the longer-term average context. According to Mcilveen (2010), climate change can be defined as climatic dissimilarities lasting for a period of time. The Intergovernmental Panel on Climate Change (IPCC), describes climate change as "any variation in climate over time, whether due to natural variability or as a result of human action" (IPCC, 2007). The IPCC definition, however, contrasts that of the United Nations Framework Convention on Climate Change (UNFCCC) (UNFCCC, 2013), where climate changes is the result of direct or indirect human activities modifying atmospheric composition over a period of time. Two main practical methods to address climate change are mitigation and adaptation (Harrison et al., 2010; UNFCCC, 2013). Modelling of climate change has become a modern tool in addressing this new phenomenon.

Analysis of climate model results seems to suggest that anthropogenic changes, particularly increases in greenhouse gases (GHG), are probably responsible for alterations in average weather conditions. Lately, global issues related to climate change are dominating environmental agenda forums as well as platforms grabbling with agricultural concerns, as shown by the upsurge of scientific articles dealing with climate change (Adams, 1989). Even if the course of greenhouse gases concentrations in the atmosphere could be forecast with confidence, future climates fate would still be indeterminate. Analysis of current climate models seems to suggest that they are not too reliable except global general circulation models (GCMs), which are large-scale mathematical depictions of the climate system (Kaiser, 1991). Forecast from GCMs on the climate system indicate that a doubling of atmospheric CO₂ will increase global average temperature from 1.5 to 4.5°C. In addition, it will modify precipitation amounts and frequency, however, other current climate models predict

an average increase of 2.8 to 5.2°C in average global temperature over the next century (Karl et al., 1990). Regardless of average global temperature increase, changes in weather is expected to impact crop and livestock production including other constituents of agricultural systems (Eitzinger et al., 2013). It is guite clear now that changes in the average global weather conditions alter rainfall patterns, evaporation levels, runoff and soil moisture storage, which as a consequence affect the growth and development of plants. In agreement with this view, it is reported that the incidence of inadequate moisture during different growth stages of plants (flowering, pollination and grain-filling) is harmful to most grain crops (e.g. cowpea, maize, soybean and wheat) (Eitzinger et al., 2013). Furthermore, extant literature shows that crop and livestock yields are directly affected by changes in climatic factors such as temperature and precipitation and the frequency and severity of extreme instances like droughts, floods, and wind storm (Adams, 1989). For instance, rice yield declines have been reported in south China as a result of climate change. Climatic change is expected to be less favourable for spring wheat and winter wheat yields in south China but opposite compared to northern region (Houxuan et al., 1993). Other studies in the east and central regions of China conclude that maize yields would decline if there is no increase in available moisture (Houxuan et al., 1993). In another interesting study conducted in China, modelling of soybean growth indicates yield expansions in the far north-east, but, notwithstanding faster growth, yield declines in most other areas (Smit & Yunlong, 1996). Clearly, the advent of climate change will significantly impact crop production directly and household food and nutrition indirectly. It is important to highlight that the climate change impact on crop yields, agricultural productivity and food security vary depending on the types of agricultural practices and methods (Watson et al., 1997). African agricultural production systems may significantly be affected by the increase in global warming (Watson et al., 1997). Developing countries i.e. India, Brazil, South Africa etc. rely more on farming for food and nutrient acquisition, and the majority of the farms are situated in areas already too hot or dry and the poor farmers are less able to adapt to changes in climate (Mendelsohn et al., 1994). The effects of global climate changes on agricultural production are likely to be small to moderate and in particular, large commercial farms are therefore more vulnerable to global warming than small household farms (Aydinalp & Cresser, 2008). In the advent of climate change, small household farmers dependent on subsistence farming are particularly vulnerable to food and nutrition insecurity. Notably, many of these at-risk populations are located in developing countries (Aydinalp & Cresser,

2008). Climate can be a significant determinant for future price trends therefore the steadiness of whole food systems may be at severe risk under climate change. It is easy to speculate that future food and crop prices will increase due to problems associated with climate change. According to Kaiser (1991), the prices for corn and soybeans will tend to be higher than for wheat and coarse grains and these price increases will be due to declines in overall production in developed countries e.g. European regions, U.S and Canada. Ultimately, these circumstances will result in food security being threatened. According to the United Nation's Food and Agriculture Organization (FAO) (2002), "food security refers to a "situation that exists when all people, at all times, have physical, social, and economic access to sufficient, safe, and nutritious food that meets their dietary needs and food preferences for an active and healthy life". Accordingly, everyone has a basic right to enough diet (FAO, 2002). All four dimensions of food security are affected by climate change i.e. food production and availability, the stability of food supplies, access to food and food consumption (IPCC, 2007). Over and above climate change, socio-economic impacts on food production, food security also depends on economic growth, changes to trade flows, stocks, and food aid policy (Prajapati et al., 2011). Climate change does not only have an impact on crop production but livestock as well. Analysis conducted exposes that the impacts of climate change differ across countries. Countries with cool environment are likely to encounter livestock losses from warmer temperatures due to the loss of beef cattle, and irrigated crops in presently hot regions such as Ethiopia and West Africa will suffer from global warming, although irrigated crops in the Nile Delta and Kenyan highlands will gain (Kurukulasuriya et al., 2011). In these instances, where global warming is increasing incredibly, farmers should consider growing crops such as cowpea which are resistance or adaptable to warm climatic conditions. Cowpeas have the ability to grow under difficult and stressful conditions in marginal areas (Ntombela, 2012). Cowpea's resilience in the different agro-ecological regions in sub-Saharan Africa is the main reason why it is the cheap number one protein source in the human diet. In an interesting field study conducted in Ghana and South Africa, with different agro-ecological climatic conditions, Belane (2011) investigated the effect of interaction of genotype and environment on plant growth and N2 fixation of 30 cowpea genotypes. The author reported significant differences in the interaction effect on cowpea biomass yield, symbiotic parameters, grain yield and nutritional contents in both leaf and grain among the studied genotypes. Furthermore, the study revealed that cowpea leaves had more minerals (3-fold of Fe) compared to spinach, which is a

common vegetable in South Africa (Belane & Dakora, 2010; Belane, 2011). Notably, Fe is among the micronutrients that are deficient in most African diets. In Africa, cowpea is an essential component of food and nutrition security (Seed World Cowpeas (SWC), 2017).

2.6 COWPEA CULTIVATION FOR FOOD SECURITY

Cowpea (Vigna unguiculata L. Walp) is an important nutritious legume grain crop that is cultivated under different agro-ecological regions around the globe. In sub-Saharan Africa, it is a staple food for most households that provides cheap and rich nutrients such as protein, minerals and vitamins (Bressani, 1985) and fodder (Tarawali et al., 2002). Cowpea could be grown as a multi-purpose crop providing high protein grain for human consumption and crop residues of high nutritive value for animal feed and as a forage crop. In addition, as a staple food crop in sub-Saharan Africa, cowpea is commonly cultivated as an intercrop with other crops such as maize and sorghum including some vegetables such as okra and spinach, and recently in some traditional farms, it is intercropped with cassava and cotton (Rusinamhodzi et al., 2006). Cowpea as a legume, can contribute to soil fertility, through its nitrogen fixing abilities. One of the advantages of using cowpea in intercropping and rotational cropping systems in Africa is because of its ability to fix a large amount of N (up to 201 kg N ha⁻¹) (Dakora et al., 1987). Cowpea's contribution in terms of nitrogen provision and other benefits to cropping systems are well documented in Africa (Dakora et al., 1987; Ofori & Stern, 1987; Dakora & Keya, 1997; Ayisi et al., 2000; Caskey et al., 2001; Bado et al., 2006; Ncube et al., 2007; Adjei-Nsiah et al., 2008).

Cowpea is an important forage crop with high yield and quality and it is adaptable to harsh environmental conditions (Ayan et al., 2012). In semi-arid tropics covering Asia, Africa, Central and South America, cowpea is a valuable food legume and component of the traditional cropping systems. Cowpea has many uses: the young leaves, green pods and green seeds are used as vegetables and dry seeds are used in various food preparations (Singh et al., 2003). Cowpea hay is a nutritious balanced feed for animals in West Africa during the dry season and has an ample purpose in feeding animals (Tarawali et al., 2002). For a higher yield and quality, cowpea can be intercropped with other crops compared to single cropping. Cowpea is a drought resistant crop that has potential to withstand future climate scenarios associated with climate change and therefore, can be cultivated as legumes with low water requirement to replace the

water intensive conventional species in forage production systems (Rao & Shahid, 2011).

In sum, cowpea has a huge potential for symbiotic N₂ fixation and fits well into traditional cropping systems in most regions in Africa. The crop matures faster than most other crops and it's a cheap alternative source of nutrients and food that bridges the hunger-gap between previous and current harvests, therefore, it can serve as a food/nutrition security crop for rural communities in the advent of climate change in sub-Sahara Africa in particular and the world in general.

2.7 CONCLUSIONS

This chapter has identified AMD as a source of pollutant for surface water sources and irrigating of crops with contaminated water such as mine water has a long-term effect on both crop productivity and soil properties. Long-term irrigation with AMD water can bring changes in the quality of soil as trace element inputs are continued over a long period of time. Heavy metal contamination in agricultural soils has become a serious threat to crop production globally. Microorganisms (e.g. *B. subtilis)* have the potential for bioremediation of badly polluted waters such as AMD water. Climate change (elevated temperature) can significantly impact a plant's physiology and the production of important economic crops such as cowpea. The next chapter present findings on cowpea seed germination as affected by elevated temperature.

CHAPTER 3: SEED GERMINATION INDICES OF COWPEA EXPOSED TO ELEVATED TEMPERATURE AND BACILLUS SUBTILIS (BD233) INOCULATION

3.1 ABSTRACT

This experiment investigated the interactive effect of Bacillus subtilis (BD233) and elevated temperature treatment on seed germination indices and plumule lengths of three cowpea genotypes (Asetanapa, Soronko and Nyira) from northern Ghana. The experimental treatments included cowpea seeds treated with *B. subtilis* (BD233) (B+) and without B. subtilis (BD233) (B-), and each treatment was incubated at six different temperature regimes (10, 25, 30, 35, 40 and 45°C) for seven days. A standard germination test (three replicates of 30 seeds per genotype per treatment) repeated twice was carried out to determine the following germination indices; germination percentage (G%), germination index (GI) and germination rate index (GRI). The interaction of *B. subtilis* (BD233) and temperature significantly (*P*<0.05) influenced the germination indices and plumule length of cowpea seedlings and the differences in genotype responses were also significant. The highest G% was for Nyira (98.6%) followed by Soronko (84.8%) seeds treated with Bacillus (B+) and without Bacillus (B-) respectively and incubated at 35°C. With or without *B. subtilis* (BD233), all three genotypes incubated at either 10, or 40 or 45°C had the lowest G% compared to the other temperatures. In general, Nyira seeds treated with B. subtilis (BD233) and Soronko seeds without the bacteria obtained the fastest germination rates (lowest GI & GRI values) compared to Asetanapa seeds treated either with or without the bacteria. Soronko seeds inoculated with B. subtilis (BD233) and incubated at 35°C for seven days had the longest plumule (1.40 cm) compared to the other genotypes under similar conditions. B. subtilis (BD233) inoculation has the potential to positively influence germination and growth of cowpea, an important candidate for food/nutrition security in Africa.

3.2 INTRODUCTION

Globally, the climate is changing rapidly and regional temperatures especially in tropical Africa are predicted to rise by 2-5°C over the next 50-100 years (IPCC, 2007). The impact of this forecasted rise in regional temperatures on vegetation requires a serious attention (Sershen et al., 2014) especially in tropical Africa where the majority of the population depends on subsistence farming for their livelihood, nutritional needs

and food/nutrition security. In particular, elevated temperatures associated with climate change have been reported to affect seed germinability (seedling viability and vigour) and seedling establishment in a number of crops (Han et al., 2009) including cowpea, which is believed to have evolved in tropical West Africa (Smartt, 1985).

Cowpea (Vigna unguiculata L. Walp) is an important crop that belongs to the family Fabaceae/Leguminosae (Padulosi & Ng, 1997). The crop is mostly cultivated because of its ability to contribute nitrogen to nutrient-poor soils in addition to providing affordable dietary protein, carbohydrate, minerals and vitamins for humans (Bressani, 1985). Cowpea grain is rich in protein (42%) and carbohydrate (57%) and the green leaves and young pods are capable of supplying up to 35% protein when consumed as vegetables (Kadam et al., 1984). The crop is cultivated in tropical and sub-tropical regions with relatively high temperatures (28-30°C). However, elevated temperatures (>35°C) can lead to the production of reactive oxygen species (ROS) that are harmful to plant seeds undergoing germination (Pukacka & Ratajczak, 2005). During seed germination, elevated temperatures due to the predicted global climate change scenarios can enhance ROS production, which can affect cell wall formation and radicle elongation (Bailly, 2004; Müller et al., 2009) and influence germination and growth of this important crop thereby posing threats to future food/nutrition security. In future, there would be changes in agricultural crop production/ food security due to climate change. High temperatures can have adverse effects on crop yields where crops suffer overall reduction in growth (Ewing, 1981). There is need to mitigate impact of elevated temperature on seed germination especially on important economic crops such as cowpea and cultivation of crops in general.

Interestingly, a ubiquitous microorganism, *Bacillus subtilis*, and temperature have been reported to stimulate seed germination in some plants (Paredes-Páliz et al., 2016). The use of *Bacillus species* in agriculture is increasing steadily as it offers an alternative to expensive synthetic chemical fertilizers. Certain strains of *Bacillus* spp. are noted to increase the growth and development of plant through several mechanisms but mainly by providing siderophores and phytohormones (auxins, cytokinins, gibberellins etc.) (Ali et al., 2009a) or enhancing nutrient uptake (solubilisation of minerals) from plant rhizosphere soils (Schwartz et al., 2013; Meng et al., 2016). A study has found that *B. subtilis* has the ability to quench soil ills to boost plant growth (Adesemoye et al., 2008). However, no studies have explored the interactive effect of *B. subtilis* and temperature on seed germination indices of cowpea,

an important crop in Africa and other regions of the world (Freire-Filho, 2011). The experiment aimed to investigate the effect of *B. subtilis* and elevated temperature on seed germination indices and plumule lengths of three cowpea genotypes (Asetanapa, Soronko and Nyira) and compare genotype responses across treatments.

3.3 MATERIALS AND METHODS

3.3.1 Seed Materials and Measurements

Seeds of three cowpea genotypes (Asetanapa, Soronko and Nyira) were obtained from the Crop Research Institute, Kumasi, Ghana. Thirty randomly selected seeds per genotype were used for measurement of their grain lengths and widths. The length and width of the seed were measured using a digital vernier caliper (Model DC–515).

3.3.2 Bacterial Strain

The bacterial strain (*Bacillus subtilis* (BD233)) used in this study was obtained from the National Collection of Bacterial Species of the Agricultural Research Council, Plant Protection Research Institute, Roodeplaat, South Africa. *Bacillus subtilis* (BD233) was cultivated in lysogeny broth at 37°C overnight and the pure cultures adjusted to 0.5 McFarland Standards with sterile distilled water before the germination experiment.

3.3.3 Germination Experiment and Experimental Design

In all experiments and for each genotype, 30 cowpea seeds were surface-sterilized by soaking in 70% ethanol for 90 s, and then in 1.5% bleach for 15 min, and then rinsed six times with sterile distilled water (Pule-Meulenberg et al., 2010). The experimental treatments included the following: 1) seeds treated with *B. subtilis* (BD233) (B+) and II) seeds not treated with *B. subtilis* (BD233) (B-) and each treatment was incubated in darkness at six different temperature regimes (10, 25, 30, 35, 40 and 45°C) for seven days. A standard seed-germination experiment arranged in a complete randomized block design (three replicates of 30 seeds per genotype per treatment) was carried out under sterile conditions using 9 cm plastic Petri dishes and Whatman No. 1 filter paper discs. The filter paper discs were all kept moistened with sterile distilled water throughout the experiment. Each set of experiments was repeated twice.
In each experiment and for each treatment, the number of germinated seeds (radicle/plumule emergence) was recorded every day for seven days and the seedling plumule length measured (cm) after seven days. Normal seedling produced was used for seed germination indices evaluation (ISTA, 2006). Germination indices including germination percentage (G%), germination index (GI) and germination rate index (GRI) were calculated from germination data according to Olisa et al. (2010) as follows:

$$G\% = \frac{No of emerged seedlings at final count}{Total number of seeds planted} \times 100$$

$$GI = \frac{\sum(Nx)(DAS)}{2}$$

 $GI = \frac{1}{Total number of seedlings that emerged at the final count}$

Where *Nx* is considered the number of seedlings that emerged on days after planting and DAS is the day after planting.

$$GRI = \frac{GI}{G\%}$$

3.3.4 Data Analysis

The collected data were subjected to analysis of variance (ANOVA) using the STATISTICA software version 10 (StatSoft Inc., Tulsa, OK, USA). Treatment means were separated using Duncan's multiple range test at p<0.05.

3.4 RESULTS AND DISCUSSION

The aim of this experiment was to evaluate the interactive effect of *B. subtilis* and elevated temperature on the seed germination indices and growth of an important leguminous grain and forage crop, cowpea. One objective was to compare the responses of the different cowpea genotypes to *B. subtilis* inoculation and exposure to different temperature regimes. Clearly, cowpea seed characteristics analysis showed that the grains were different in seed coat colour and size. Nyira seeds were reddish-brown in colour and the smallest in size (5.87 mm (length) and 4.33 mm (width)) compared to the brown and largest Asetenapa seeds (8.33 mm (length) and 5.73 mm (width)) (Table 3.1). The differences in seed coat colour can be attributed to types of anthocyanin and other flavonoids synthesized by the plant (Holton & Cornish, 1995). Anthocyanin genes are found in almost all plants and are known to control the pigmentation of multiple plant parts (Chandler et al., 1989) including the seed coat colour in cowpea, which has an influence on consumer preference. In a classical

germination experiment in which seed sizes of cowpea genotypes from different African countries were compared in relation to germination, Craufurd et al. (1996) found a correlation between seed size and rate of seed germination. The authors reported that the smallest-seeded cowpea cultivar, TVu 946, had the fastest seed germination rate compared to the largest-seeded cultivar, TVu 8342, which had the slowest germination rate (Craufurd et al., 1996). Similar to this report, findings in this experiment revealed that the largest-seed genotype, Asetanapa, had a lower germination percentage/index and rate index when compared to the smallest-seeded genotype, Nyira (Tables 3.2 & 3.3, Figure 3.1).

Genotype	Length (mm)	Width (mm)	Seed Coat Colour
Asetanapa	8.33±0.2a	5.73±0.2a	brown
Soronko	7.20±0.2b	5.07±0.1b	white
Nyira	5.87±0.2c	4.33±0.2c	Reddish-brown

Table 3. 1: Cowpea genotype seed measurements (n=30).

Values (M±S.E.) followed by dissimilar letters within a column are significantly different at p≤0.05.

Another objective was to evaluate the interactive effect of *B. subtilis* (BD233) inoculation and incubation temperature on the germination indices (G%, GI, & GRI) of three cowpea genotypes obtained from Ghana in tropical West Africa. According to findings in this experiment, the interaction of *B. subtilis* (BD233) and temperature significantly (p<0.05) influenced the germination indices of all three cowpea genotypes. The highest G% was obtained for Nyira seeds (98.6%) treated with B. subtilis and incubated at an elevated temperature of 35°C (Table 3.2). Elevated temperature significantly affected seed germination percentages as all three cowpea genotypes incubated at 10°C and supra temperature (45°C) had the lowest seed germination percentages (0%) compared to those incubated at temperatures of 25°C, 30°C and 35°C (Table 3.2). The cardinal germination temperatures of cowpea seeds are 25°C-40°C but in this experiment, treating seeds of cowpea genotypes with B. subtilis (BD233) showed enhanced germination percentages when incubated at a temperature range of 25-35°C. For example, Nyira seeds treated with B. subtilis (BD233) and incubated at 35°C had a significantly higher germination percentage (98.6%) compared to those incubated at an ambient temperature of 30°C (50.3%). For both Nyira and Soronko genotypes, the temperature range for optimum seed germination percentage was 25-35°C regardless of the treatment, but all three

genotypes showed 0% seed germination at 10°C in this experiment (Table 3.2). Similar to this finding, Covell et al. (1986) and Craufurd et al. (1996) reported lack of seed germination of some cowpea genotypes from West Africa at temperatures below 10°C and maximum seed germination at temperatures of 35.8°C and 35°C respectively. In another study, Motsa et al. (2015) reported optimum seed germination of cowpea at 36°C with a cardinal temperature of 30-36°C. The noted differences in optimum and cardinal temperatures of these studies when compared to the present study findings could be attributed to genetic variability among the seeds used in each of these studies (Mohammed et al., 1988) or to seed accession differences in latitude (Saeidnejad et al., 2012).

In this experiment, treating seeds of both Nyira and Soronko with *B. subtilis* (BD233) enhanced germination percentage (98.6%) when compared to those not treated with the bacteria (50.8%) and incubated at optimum elevated temperature (30 and 35°C) (Table 3.2). In agreement with findings in this experiment, Manjula and Podile (2001) reported that *Bacillus subtilis* AF 1 in peat formulations significantly stimulated percentage seed germination of two leguminous crops (groundnut and pigeon pea). Similarly, Niranjan et al. (2003) reported a positive influence of two strains of *Bacillus* on the germination and growth of pearl millet.

	Germination (%)								
	Asete	enapa	Soro	onko	Ny	Nyira			
Temp. (°C)	(B+)	(B-)	(B+)	(B-)	(B+)	(B-)			
0	0.0±0.0dA	0.0±0.0bA	0.0±0.0cA	0.0±0.0cA	0.0±0.0dA	0.0±0.0dA			
25	14.1±0.9bC	5.3±0.3aC	80.0±6.9aA	81.1±4.4aA	37.8±2.9cB	12.2±1.2cC			
30	30.2±2.4aC	6.2±0.7aD	78.4±7.3aA	84.4±2.1aA	50.3±5.5bB	38.9±2.9bC			
35	9.4±0.8cC	6.7±0.5aC	78.7±5.8aA	84.8±2.2aA	98.6±1.4aA	50.8±5.4aB			
40	0.0±0.0dC	0.0±0.0bC	19.8±0.1bA	8.9±0.1bB	0.0±0.0dC	0.0±0.0dC			
45	0.0±0.0dA	0.0±0.0bA	0.0±0.0cA	0.0±0.0cA	0.0±0.0dA	0.0±0.0dA			

Table 3. 2: Seed germination percentage (G%) of cowpea genotypes inoculated with *B. subtilis* (B+) or without *B. subtilis* (B-) and incubated at different temperatures.

Values (M±S.E.) followed by dissimilar letters (in lower case) within a column for each genotype are significantly different at $p \le 0.05$. Values (M±S.E.) followed by dissimilar letters (in upper case) within a row for each genotype are significantly different at $p \le 0.05$.

Similarly, the rate of seed germination was affected by inoculation with B. subtilis (BD233) as significant differences in other germination indices were revealed after four days' incubation at 35°C temperature (Table 3.3). Inoculation with *B. subtilis* (BD233) enhanced the germination rate of Soronko seeds compared to Nyira. The smaller the GI and GRI values of seeds, the faster is their rate of germination (Adetumbi et al., 2011). Seeds of Nyira inoculated with *B. subtilis* (BD233) and incubated at 35°C for four days showed a higher G% (67.3%) compared to Soronko (55.5%) and Asetenapa (7.7%) under the same conditions (Table 3.3). This finding implies that Nyira seeds (smallest in size) germinated at a faster rate compared to Soronko seeds. Similar to this finding, the germination indices of two elite cowpea genotypes (Ife BPC and Ife Brown) were affected significantly by storage condition (Adetumbi et al., 2011). In their study, Ife BPC had smaller GI and GRI values with the fastest germination rate but a lower final germination percentage compared to Ife Brown, which also indicated differences in cultivar responses. In this experiment, differences in genotype responses to B. subtilis (BD233) inoculation and incubation temperature were also noted.

	G(G(%)		91	G	RI
Cultivar	(B+)	(B-)	(B+)	(B-)	(B+)	(B-)
Asetenapa	7.7c	5.0c	2.23c	0.04b	0.06a	0.06a
Soronko	55.5b	44.4a	3.08a	0.03b	0.06a	0.06a
Nyira	67.3a	38.8b	2.93b	0.14a	0.03b	0.07a

Table 3. 3: Seed germination indices of cowpea genotypes inoculated with *B. subtilis* (B+) or without *B. subtilis* (B-) and incubated for four days at 35°C temperature.

Values (M \pm S.E.) followed by dissimilar letters within a column are significantly different at p≤0.05.

Interestingly, genotypic differences were noted in the cumulative seed germination percentages for cowpea inoculated with *B. subtilis* (BD233) over the experimental period. There were significant differences in rate of achieving over 50% seed germination amongst the three genotypes at all incubation temperatures (Figure 3.1). At both 25°C and 30°C, only Soronko seeds inoculated with *B. subtilis* (BD233) attained above 50% seed germination after four days' incubation compared to seeds of the other two genotypes. At 35°C, both Soronko and Nyira seeds achieved 50% germination at four days of incubation, and at 40°C, the highest seed germination was obtained for Soronko but it was below 50% at seven days of incubation (Figure 3.1).

Similarly, the interaction of *B. subtilis* and temperature also significantly (p<0.05) influenced the growth of cowpea seedlings. Soronko seeds inoculated with *B. subtilis* (BD233) and incubated at 35°C for seven days had the longest seedling plumule (1.40 cm) compared to seedlings of the other genotypes under similar conditions. At 25°C, with or without *B. subtilis* (BD233) inoculation, Soronko seedling plumule was significantly longer than those of the other genotypes. Inoculation with *B. subtilis* (BD233) markedly increased the plumule length of the test genotypes when compared to those evaluated at different temperature regimes without the bacteria (Table 3.4). For example, Asetenape inoculated with the bacteria showed greater plumule length at 30°C while Soronko at 35 and 40°C, and Nyira at 25°C, 30°C, and 35°C (Table 3.4). The enhanced plumule length of Soronko and Nyira seeds inoculated with the bacteria is a clear indication that *B. subtilis* (BD233) has the potential to influence growth of cowpea even at a higher temperature. In agreement, *Bacillus* sp. was also reported to enhance the growth and yield of French beans (Saxena et al. 2013), sorghum



(Prathibha & Siddalingeshwara, 2013), groundnut and pigeon pea (Manjula & Podile, 2001), soybean (Bai et al., 2003) and pearl millet (Niranjan et al., 2013).

Figure 3.1: Cumulative germination percentages (G%) of cowpea genotypes inoculated with *B. subtilis* (BD233) and incubated at different temperatures (25, 30, 35 and 40°C).

	Plumule length (cm)									
	Asete	enapa	Sorc	onko	Ny	Nyira				
Temp. (°C)	(B+)	(B-)	(B+)	(B-)	(B+)	(B-)				
25	0.13±0.0bC	0.08±0.0aC	0.92±0.1bA	1.11±0.1aA	0.56±0.0bB	0.17±0.0bC				
30	0.48±0.0aB	0.07±0.0aC	0.93±0.1bA	1.08±0.1aA	0.61±0.0bB	0.15±0.0bC				
35	0.08±0.0cD	0.05±0.0bD	1.40±0.1aA	1.14±0.1aB	1.08±0.1a <mark>B</mark>	0.56±0.0aC				
40	0.00±0.0cC	0.00±0.0cC	0.42±0.0cA	0.24±0.0bB	0.00±0.0cC	0.00±0.0cC				

Table 3. 4: Seedling plumule length of cowpea genotypes inoculated with *B. subtilis* (B+) or without *B. subtilis* (B-) and incubated at different temperatures.

Values (M±S.E.) followed by dissimilar letters (in lower case) within a column for each genotype are significantly different at $p \le 0.05$. Values (M±S.E.) followed by dissimilar letters (in upper case) within a row for each genotype are significantly different at $p \le 0.05$.

3.5 CONCLUSIONS

The interaction of *B. subtilis* (BD233) and temperature significantly influenced the germination indices and plumule lengths of all three cowpea genotypes included in this experiment. Findings in this experiment show that the cardinal temperature for the three West African cowpea genotypes was 25-35°C and the seed germination percentage ranged from 78.4 to 98.6%. Clearly, inoculation with *B. subtilis* (BD233) significantly influenced seed germination and growth of all three cowpea genotypes at elevated temperatures. *B. subtilis* (BD233) inoculation has potential to positively influence the germination and growth of cowpea, an important candidate for food/nutrition security in Africa under future climate change scenarios.

CHAPTER 4: GROWTH AND NUTRITIONAL RESPONSES OF COWPEA (Cv. Soronko) TO SHORT-TERM ELEVATED TEMPERATURE

4.1 ABSTRACT

An evaluation of the effect of elevated temperature on the growth and nutritional quality of cowpea was undertaken in this experiment. Surface sterilized seeds of cowpea (Cv. Soronko) were germinated in pots in the glasshouse. At different growth stages (pre-flowering (40 DAP), flowering (90 DAP) and post-flowering (123 DAP)), plants were incubated in growth chambers set at three different temperature regimes (25, 30, & 35°C) for a period of seven days. Whole plant biomass (fresh and dry weight), and shoot carbon and crude protein contents were significantly affected with increase in temperature at all three stages of the plants' growth. At an elevated temperature (35°C), the pre-flowering and flowering stages of cowpea where most affected compared to control (25°C). The results suggest that the pre-flowering and flowering stages of cowpea (20-35°C).

4.2 INTRODUCTION

Cowpea (Vigna unguiculata L. Walp) is an important nutritious legume cultivated in most regions in Africa including South Africa. In South Africa, rural farmers cultivate cowpeas for subsistence and source of income and nutrients but also to a lesser extent as feed for livestock (Kritzinger, 2004), whereas the crop is utilised largely for other purposes in other countries in Central and West Africa (Singh et al. 2002). For example, in West Africa, the leaves are eaten as vegetables. The seeds can be cooked and eaten with as a meal. When the young green leaves of cowpea are eaten as a vegetable, it can provide between 27.1 and 34.7% protein to humans and animals (Jasper & Norman, 1983; Ahenkora et al., 1998). Mature seeds contain per edible portion (100 g): 56-66 g carbohydrate, 22-24 g protein, 11 g water, 5.9-7.3 g crude fibre, 3.4–3.9 g ash, 1.3–1.5 g fat, and minerals such P, Ca, and Fe (0.146 g, 0.104– 0.076 g, & 0.005 g respectively) (Asante et al., 2006). The edible grain of cowpea is suggested to contain a protein range of 22-25% (Oyenuga 1959; Platt, 1962), and carbohydrates and ash contents of 56.8% and 3.6%, respectively (Platt, 1962). A research conducted on the nutritional value of the cowpea leaves and immature pods, recommends that the leaves have similar nutrition value to black nightshade and sweet potatoes leaves, while the green pods have less anti-nutritional factors than the

dried seeds (Gonçalves et al., 2016). Clearly, the nutritional benefit associated with the grain and leaves of cowpea is not well documented, however, the crop is a cheap source of nutrients for rural communities in South Africa. This legume crop has the advantage of growing in poor-nutrient soils in marginal areas in the country because of its ability to perform symbiotic nitrogen fixation with microorganisms (rhizobacteria) (Belane, 2011). In addition, the crop is tolerant to heat and drought associated with climate change in marginal areas, and therefore can be considered as a climatechange friendly crop (Awika & Duodu, 2017).

Globally, changes in climate can alter rainfall patterns, evaporation, runoff and soil moisture storage in most regions including areas under agricultural cultivation, which as a result affects the growth and development of crops. It is reported that the occurrence of insufficient moisture during flowering, pollination and grain-filling is harmful to most crops (Eitzinger et al., 2010). In addition, increases in average temperatures or short-term temperature increases causing high-temperature stress on plants are associated with changes in climate change. Changes in temperature are noted to affect the physiological processes and metabolite profile of plants including nutritional status, which ultimately affects their growth and development. Earlier studies have reported on the sensitivity of the different growth stages of plants to climatic conditions (Prasad et al., 2002; Hatfield et al., 2008; Hatfield & Prueger, 2015) including legumes. Agricultural systems in the tropics are expected to be impacted by drastic short-term changes in temperature and soil moisture. In particular, African agricultural production may significantly be affected by the increase in global warming associated with climate change (Watson et al., 1997). In developing sub-Saharan countries such as South Africa, majority of cowpea cultivation is located in places that are already too hot or dry, and poor farmers in rural areas, are less able to adapt (Mendelsohn et al., 1994). In Sub-Saharan Africa, cowpea has received more research attention leading to breeding for earliness, high grain yield, and pest and disease resistance (Singh & Ntare, 1985). However, there is need for rural farmers to cultivate cowpea genotypes that are tolerant to future elevated temperatures. In addition, literature is scarce on the effect of elevated temperature on the crop's nutrition especially at different growth stages. Therefore, the aim of the experiment was to evaluate the effect of short-term elevated temperature on the nutritional quality of cowpea at different growth stages.

4.3 MATERIALS AND METHODS

4.3.1 Plant Material

Among the three cowpea genotypes employed in this study, the Soronko genotype was noted to have the highest germination percentage and longest plumule length at elevated temperature (35°C) and was therefore selected for this experiment. The seeds were obtained as described in Chapter 3, section 3.31.

4.3.2 Plant Growth Media and Experimental Conditions

Acid washed sand was obtained from Masstores (Pty) Ltd, Sunnyhill, Johannesburg, South Africa. The acid sand was thoroughly washed using tap water to remove nutrients. One litre planting pots (14-inch) were used for the experiment. Surfacedsterilized cowpea (cv Soronko) seeds (3) were germinated in pots containing acid washed sand and the emergent seedlings were thinned to one per pot. The seedlings were grown in a glasshouse set at a day/night temperature regime of 25/18°C with a supplementary light intensity of 300 µmol m⁻²s⁻¹ and an overhead irrigation system was used with each plant receiving 63 mL of water a day split into three regimes of 5 mins each. The experiment was conducted at the University of South Africa (Unisa) Florida campus, Gauteng Province (26° 10′ 30″ S, 27° 55′ 22.8″ E).

4.3.3 Fertilisation of Cowpea

Seedlings of cowpea in pots were grown with modified ¹/₄-strength N-free solution as a source of nutrients (Broughton & Dilworth, 1970). Stock solutions of the N-free nutrient was prepared as described by Broughton and Dilworth (1970) and details of salts used are shown in Table 4.1 below.

Stock	Form	Source/Element	Mass/Litre	Concentration
Solutions				
Α	CaCl ₂ .2H ₂ O	Ca	294.1	2.0 M
В	KH ₂ PO ₄	Р	136.1	1.0 M
С	Fe-citrate	Fe	607	20.0 mM
	MgSO ₄ .7H ₂ O	Mg	123.3	0.5 M
	K ₂ SO ₄	K	87.0	0.5 M
	MnSO ₄ .2H ₂ O	Mn	0.338	2.0 mM
D	H ₃ BO ₃	В	0.247	4.0 mM
	ZnSO4.7H2O	Zn	0.288	1.0 mM
	CuSO ₄ .5H ₂ O	Cu	0.100	0.4 mM
	CoSO ₄ .7H ₂ O	Co	0.056	0.2 mM
	NaMoO ₂ .2H ₂ O	Мо	0.048	0.2 mM

Table 4. 1: N-free plant nutrients and their concentrations in stock solutions.

For each 10 L of full strength stock solution, 5.0 mL of each of solutions (A-D) was taken and added to 5 L of water, then diluted to 10 L. One N NaOH was used to adjust the pH to 6.6-6.8 (Broughton & Dilworth, 1970). The seedlings were irrigated with the N-free nutrients solutions twice during the seven days in growth chambers (Nüve test cabinet TK 120, China) (Figure 4.1).

Commercial inoculants for cowpea were obtained from STIMUPLANTS CC in Pretoria. Specifically, a packet of the powdered inoculant with *Bradyrhizobium spp* CB756 strains was used for inoculation of cowpea seedlings. About 2 g of the powder inoculant was mixed with 500 mL of distilled water and a sticker and applied as a slurry to cowpea seeds before planting. Two weeks after seedling establishment the same mixture of inoculant was applied close to the root of seedlings to ensure inoculation of each plant.

4.3.4 Experimental Treatments: Biomass Determination and Nutrient Analysis

4.3.4.1 Cowpea growth and nutritional responses to elevated temperature

For this experiment, three seedlings of cowpea (cv Soronko) were exposed to three temperature regimes (25°C (control), 30 & 35°C (elevated)) as treatments for a period of seven days. For each of the above-mentioned treatments, three seedlings of similar phenology were randomly selected from seedlings grown in the glasshouse at three different growth and developmental stages (pre-flowering (40 days after planting (DAP)), flowering (90 DAP) and post-flowering (123 DAP)). The plants were incubated in three separate growth chambers set at the different temperatures for a one-week period (Figure 4.1). In each growth chamber, the humidity was set and maintained at 70% and the lights were on (00h00-12h00) and off (12h01-23h59) for 12 h for seven days. Seedlings were irrigated with tap water every 24 h for seven days and fertilized with modified ¼-strength N-free solution twice in seven days. The three different stages of the plant's development (pre-flowering, flowering and post-flowering) exposed to the different temperature regimes were assessed after seven days of incubation for the effect of temperature on the growth and nutritional quality of cowpea.

Throughout the experiment there were no incidences of plant diseases and pest and the plants were healthy before and during the 7-day incubation period.



Figure 4.1. Growth chamber- Nüve TK Series Test Cabinets (Nüve test cabinet (model TK 120))

4.3.4.2 Plant biomass

For each treatment, the fresh and dry weights of the whole plant (shoot and root) were determined using a weighing balance. For whole plant fresh weight (FW) determination, after seven days' incubation all three plants were thoroughly washed with tap water and immediately dried using paper towels before weighting. For dry weight (DW) determination, the harvested plants were oven-dried at 60°C for 72 h to constant weight and weighed. The plant biomass (whole plant fresh and dry weights) was determined for each growth stage exposed to the treatments.

4.3.5 Determination of Nitrogen (Crude Protein) and Carbon

4.3.5.1 Sample preparation

For this experiment, finely ground oven-dried shoot of plants exposed to the three different temperature regimes (treatments) were used. Sample grinding was done using a mechanical mill (Knifetec 1095 Sample Mill) and passed through an 80-mesh sieve to obtain uniform particles. Samples were uniform powder finely ground and homogenized for repeatable and consistent results. Typically, samples should be ground to a fineness of <0.5 mm. About 0.1-0.5 g of sample was weighed for each sample analysis.

4.3.5.2 Sample analysis

Determination of nitrogen/protein and carbon in samples was done using LECO Trumac CN analyser (832 series, USA) (Figure 4.2). Approximately 0.3 g of plant sample was weighed into a crucible transferred to a suitable position of the autoloader. These steps were repeated for each sample to be analysed.



Figure 4.2. LECO Trumac C/N analyser

Before each sample analysis, pre-weighed homogeneous samples were placed into a large ceramic boat and loaded into the purge chamber located in the front of the horizontal ceramic high-temperature furnace. After the entrained atmospheric gas was purged from the sample, the ceramic boat was introduced into the furnace regulated at a temperature of 1100 to 1450°C. Complete oxidation of the macro sample was ensured by a pure oxygen environment within the furnace, with additional oxygen being directed onto the sample via a ceramic lance. The ceramic boat and all ash from the sample were removed from the furnace at the end of combustion, leaving the furnace free of ash build-up. For each treatment applied to the three different growth stages, three replicate shoot samples were analysed for the concentration of its crude protein/nitrogen and carbon contents.

4.3.6 Data Analysis

Data on growth parameters (whole plant biomass (FW & DW)) and nutritional contents (CN) of cowpea shoot were analyzed using one-way ANOVA analysis. All parameters and concentration measurements were tested at p<0.05 significance level and the Duncan multiple range test was used for separation between treatment means.

Statistica v. 10, StatSoft (StatSoft Inc., Tulsa, OK, USA)) was used for all statistical analysis.

4.4 RESULTS AND DISCUSSION

4.4.1 Cowpea Growth Response to Short-term Elevated Temperature

In this experiment, short-term elevated temperature significantly (p<0.05) affected all three growth stages of cowpea (cv Soronko) incubated in growth chambers (Table 4.2). For each cowpea growth stage exposed to a 7-day elevated temperature (30-35°C), the whole plant (stem + leaf + root) biomass decreased significantly compared to control (25°C). For example, at 40 DAP (pre-flowering stage), in comparison with control, the whole plant fresh and dry weights of cowpea exposed to elevated temperature (35°C) was reduced by 42 and 29%, respectively, and at 30°C, cowpea whole plant fresh and dry weights reduced by 30 and 52% respectively (Table 4.2). At 90 DAP (flowering/anthesis stage), in comparison with control, the cowpea whole plant fresh and dry weights exposed to elevated temperature (35°C) was reduced by 47% and 63% respectively and at 30°C, the fresh and dry weights reduced by 31% and 60% respectively (Table 4.2). Similarly, at 123 DAP (post-flowering stage) the growth response of cowpea was affected by a higher temperature (30°C) compared with control. The whole plant fresh and dry weights of cowpea reduced by 31% and 60% respectively when exposed to a high temperature (30°C) compared with control, however, no significant difference in whole plant biomass was noted between elevated temperature (35%) and control (25°C) (Table 4.2).

			Whole plan	t biomass (g/p	olant)	
Pre-flowering		Flow	vering	Post-fl	owering	
	(40	DAP)	(90 DAP)		(123 DAP)	
Temp (°C)	FW	DW	FW	DW	FW	DW
35	5.6±0.8b	3.2±0.5b	47.3±9.3c	6.3±1.4b	92.6±8.8a	18.6±4.8a
30	6.7±1.2b	2.3±0.9b	62.0±1.3b	7.0±2.9b	81.3±8.8b	14.6±2.7b
25	9.6±2.4a	4.8±1.6a	89.6±8.6a	17.3±4.3a	93.6±34.8a	18.3±6.3a

Table 4. 2: Effects of temperature on the plant biomass (g/plant) (fresh weight (FW) and dry weight (DW)) at different growth stages of cowpea grown in a growth chamber.

Values (M \pm S.E. (n=3)) followed by similar letters in a column are not significantly different at p≤0.05.

As several authors have reported, plant growth rate throughout their life cycle is affected by temperature extremes (Prasad et al., 2002; Hatfield et al., 2008; Hatfield & Prueger, 2015; Kondache et al., 2018) and the range differs among species (Lafta & Lorenzen, 1995). Across all plant species, the reproductive growth stage is more sensitive to temperature extreme and pollination is one of the most susceptible phenological stages (Hatfield & Prueger, 2015). In general, in the plant's life cycle, anthesis and early grain filling stages are more sensitive to heat shock events for most crops (Stone & Nicolas, 1994; Barlow et al., 2015). In this experiment, the preflowering and flowering stages of cowpea (cv Soronko) were more affected by temperature extremes compared to post-flowering. At 90 DAP (flowering/anthesis), plants exposed to elevated temperature (35°C) showed the highest reduction in whole plant biomass (47% and 63% in FW & DW respectively) compared with those exposed to the same temperature at other growth stages (42 and 29% at pre-flowering and 1% and 0% at post-flowering). A recent study showed that heat stress impacted the reproductive stage; the flower bud initiation, decreases the number and size of flowers, deformed floral organs resulting in loss of flowers and young pods therefore reduction in seed yield of legumes (Sita et al., 2017). Similar to this finding, the vegetative and reproductive growth stages of maize development were most affected by exposure to elevated temperature (30-35°C), and in particular, because of decreased pollen viability (Hatfield et al., 2008), the grain yield was reduced by 80-90% compared to control (Hatfield & Prueger, 2015). In another related study, high temperature (31/29°C day/night) in growth chambers significantly reduced plant biomass of two potato genotypes (total dry weights reduced by 44% and 72% in cv Norchip and cv Up-todate, respectively) and the tuber growth was more affected compared to shoot growth (Lafta & Lorenzen, 1995). Similarly, the growth stages of wheat plant were affected differently when exposed to heat shocks over short periods. Grain number reduction was highest when heat shock was applied two days prior to anthesis but was lower by two levels when applied two days after anthesis. The grain number was reduced significantly by 12-22% in five individual genotypes of wheat when exposed to heat shock 10-30 days after anthesis compared to no significant reduction in the remaining 75 genotypes tested (Stone & Nicolas, 1995). The insignificant change in whole plant biomass at post-flowering when the control was compared with exposed 35°C (Table 4.2) would suggest that the plant has the capacity to acclimate to short-term heat events (Barlow et al., 2015). Heat stress tolerance mechanisms in plants has been linked to production of heat shock proteins and in wheat plants, heat shock was

associated to reduction of grain yield and translocation of photosynthates to grains as well as starch synthesis (Acevedo et al., 2002; Barlow et al., 2015).

4.4.2 Effect of Short-term Elevated Temperature on Cowpea Crude Protein and Carbon Content

Similarly, short-term elevated (30-35°C) temperature significantly (p<0.05) affected the nutritional quality of cowpea at different growth stages. The crude protein content (Nitrogen value (%)) in shoot of cowpea (cv Soronko) was significantly affected compared to control at both the pre-flowering and flowering stages after a seven-day period incubation in growth chambers (Figure 4.3).



Figure 4.3: Effect of short-term elevated temperature on the nutritional content (Nitrogen value (%)) of cowpea cv Soronko shoot at different growth stages.

At 40 DAP (pre-flowering), the shoot crude protein content (5.59 N%) of cowpea cv Soronko exposed to short-term elevated temperature (35°C) was increased significantly compared to control (3.77 N%). In contrast, at the flowering stage (90 DAP), elevated temperature (35°C) significantly reduced the crude protein content (1.77%) of the shoot compared to control (5.59%). Interestingly, at 30°C, the nitrogen level (%) was significantly higher in the shoots compared to control during the flowering stage. However, at post-flowering (123 DAP), elevated temperatures (30°C & 35°C) compared to control had no significant effect on the crude protein content in shoots of cowpea cv Soronko plants (Figure 4.3). Clearly, this finding further showed the sensitivity to elevated temperature of the flowering stage in cowpea plants' growth and development. During the flowering stage, the crude protein content in shoots was drastically increased at 30°C and reduced at 35°C compared to control. In agreement with this finding, short-term/abrupt heat stress reduced the total protein content, nutrient-uptake levels and assimilation proteins in roots of tomato plants compared to control (Giri et al., 2017) especially at the flowering stage.

Similarly, the carbon content in shoot of cowpea (cv Soronko) at both the pre-flowering and flowering stages were affected by short-term elevated temperature compared to control. At the pre-flowering stage, the carbon content in shoot of cowpea (cv Soronko) was increased significantly at 30°C (16.7%) compared to control (15%), however, at both the flowering and post-flowering stages no significant changes in shoot carbon contents were noted but there was a slight reduction and increase at the flowering stage and post-flowering (respectively) compared to control (Figure 4.4).



Figure 4.4: Effect of short-term elevated temperature on the nutritional content (Carbon value (%)) of cowpea cv Soronko shoot at different growth stages.



Figure 4.5: Effect of short-term elevated temperature on the nutritional content (C/N ratio) of cowpea cv Soronko shoot at different growth stages.

Comparison of the C/N ratios in shoots of cowpea cv Soronko showed that short-term elevated temperature similarly affected accumulation of nutrients during the different growth stages. For example, at the pre-flowering growth stage, elevated temperature (35°C) slightly reduced the C/N ratio compared to control and at the flowering stage, the C/N ratio was significantly increased at 35°C (22.1%) compared to control (11.8%). The C/N ratio in shoots of cowpea cv Soronko exposed to short-term elevated temperature showed significant differences only at the flowering growth stage (Figure 4.5). During both the pre-and post-flowering stages, there were no significant changes in the C/N ratio in shoot of cowpea cv Soronko incubated at elevated temperature compared to control. Clearly, the CN ratio results further demonstrated the susceptibility of the flowering stage to elevated temperature in the tested crop.

In this experiment, the susceptibility of the flowering stage (90 DAP) to short-term elevated temperature (Figures 4.3-4.5) noted in changes in cowpea cv Soronko's shoot crude protein and carbon content has been demonstrated and it has implications for the crop's grain nutritional quality and yield. At anthesis, the plant's pollination process can be impacted by elevated temperature, which can result in changes in the number of grains (Barlow et al., 2015). Heat shock is reported to impact mostly grain yield during anthesis to early grain filling in wheat crop (Barlow et al., 2015).

Although nutrients in cowpea shoots were determined in this experiment, notably, elevated temperature is reported to negatively affect the nutrient sink-source relationship between the root and shoot (Huang et al., 2012), which implies that decreases in root nutrients will reduce nutrient transfers to shoot. If the growth and mass of root is affected by elevated temperature as reported by many authors (Rennenber et al., 2006; Wahid et al 2007; Huang et al., 2012; Heckathorn et al., 2014), the growth of above-ground parts (shoot) will also be affected because the supply of water and nutrients will be affected (Hao et al., 2012; Giri et al., 2017).

4.5 CONCLUSIONS

In this experiment, exposure to short-term (seven days) elevated temperature (35°C) significantly affected the growth and nutritional quality of cowpea at the pre-flowering (40 DAP) and flowering growth stages (90 DAP). Compared to other growth stages, the whole plant biomass (Table 4.2) and nutrient contents (Figures 4.3-4.5) of cowpea

at the flowering stage (125 DAP) were most affected by elevated temperature. Rural farmers cultivating cowpea should adopt cropping management strategies for the protection of the flowering stage in the crop's growth cycle especially during periods of changes in climatic conditions (temperature extremes over short periods) under climate change scenarios.

CHAPTER 5: INTERACTION EFFECT OF BACILLUS SUBTILIS INOCULATION AND MINE WATER IRRIGATION ON COWPEA'S GROWTH, PHYSIOLOGY AND NUTIRIONAL QUALITY

5.1 ABSTRACT

The aim of study was to evaluate the interaction of *Bacillus* inoculation and acid mine water irrigation on the growth, physiology and nutritional quality of cowpea. Three cowpea genotypes (Asetenapa, Soronko & Nyira) were exposed to four treatments namely i) Inoculation with *B. subtilis* (BD233) (B+) and irrigation with mine water (75% AMD) (+), ii) Inoculation *B. subtilis* (BD233) inoculation (B+) and no mine water irrigation (-), iii) No *B. subtilis* (BD233) inoculation (B-) and irrigation with mine water (+) and iv) No inoculation with *B. subtilis* (BD233) (B-) and no irrigation with mine water (-) (control). A factorial (three factors: genotype x *B. subtilis* inoculation x AMD irrigation) experiment was conducted using a randomized complete block design with three replications. The results show that interaction of *Bacillus* inoculation and acid mine water irrigation and yield of the tested cowpea genotypes and acid mine water irrigation negatively influenced the growth and nutritional quality of cowpea grains. Genotype responses to the treatments were significantly different.

5.2 INTRODUCTION

Rural communities rely on vegetables and legume crops such as cowpea to meet their mineral and protein requirements. Cowpea is deliberated as the second most important researched food grain legume in sub-Sahara African and constitutes a cheap source of plant protein for humans. As a food, the fresh leaves, seeds and immature pods of cowpea are consumed as vegetables. Cowpea leaves and seeds provide a rich source of proteins and calories, minerals and vitamins. In semi-arid regions of Africa, cowpea is widely grown and eaten in many households. The crop's genetic diversity enables its adaptation to different soil ecologies throughout the continent (Ehlers & Hall, 1997; Singh et al., 2003). It matures faster than most crops and therefore, can fill the hunger gap between cropping seasons, especially in rural communities where farmers are dependent on subsistence farming for their livelihoods. In West African countries that grow cowpeas as a major food crop, this

complements the main cereal diet (Langyintuo et al., 2003). The crop has the potential to serve as a food and nutrition security crop in sub-Sahara Africa.

In general, symbiotic legumes, including cowpea, do not only provide plant protein but also have the ability to supply adequate amounts of macro/micronutrients through their capacity to take up and concentrate minerals in their tissues (Belane, 2011). In addition, the crop has a unique ability to fix atmospheric N in symbiosis with bacteria (*Rhizobium*) thus, improving soil N fertility, which is important for system productivity and therefore fits well in the African traditional cropping systems.

In South Africa, cowpea cultivation has the potential to play a role in mitigating the threat posed by climate change on household food and nutrition security especially for poor-resource/smallholder farmers in rural communities. However, as a water-scarce country, for example, in Gauteng Province most surface waters are contaminated with AMD and there is a dire need to either clean/bio-remediate or use acid mine water for the irrigation of crops. Poor-resource farmers cannot afford expensive technologies to decontaminate or bio-remediate acid mine water discharging into freshwater sources used for irrigation of crops in their communities. In the advent of water scarcity during future climate change scenarios, it is expected that farmers will turn into wastewater/acid mine water as water source for irrigation of crops especially by resource-poor farmers in rural communities (FAO, 2010). Interestingly, the use of heavy metal resistant bacteria including the *Bacillus* consortium (Singh et al., 2014) having plant growth promotion traits (Román-Ponce et al., 2017) have been reported as eco-friendly and less cost-effective technologies for bioremediation of heavy metal polluted soils (Khan et al., 2009).

Bacillus subtilis is reported to enhance plant growth and also bio-remediate polluted soils. For example, *Bacillus* PZ3 was reported to reduce Cr in pea plants when inoculated with the bacteria in chromium amended soils and also enhanced the plant's growth, nodulation, leghaemoglobin, chlorophyll and protein content compared to control plants (Wani & Zainab, 2016). *Bacillus* spp. and other microorganisms are known to produce compounds such as siderophores and indol-3-acetic acid (IAA), which enhances the plant's growth and resistance to some heavy metals (Román-Ponce et al., 2017). However, no study has assessed the interactive effect of inoculating cowpea with any *Bacillus subtilis* strain and irrigation with acid mine water under field or glasshouse conditions. Therefore, the aim of study was to evaluate the

44

interaction of *Bacillus* inoculation and mine water irrigation on the growth, physiology and nutritional quality of cowpea. The first objective was to assess the growth, nodulation, chlorophyll content and stomatal conductance of cowpea genotypes inoculated with *B. subtilis* and irrigated with acid mine water. The second objective was to assess the nutritional contents of grains and shoots of three cowpea genotypes irrigated with acid mine water and exposed to *B. subtilis* inoculation under glasshouse conditions.

5.3 MATERIALS AND METHODS

5.3.1 Plant Material

Seeds of three cowpea genotypes (Asetanapa, Soronko and Nyira) were obtained from the Crop Research Institute, Kumasi, Ghana. The seeds were stored in storage facilities in Eureka Building, Florida campus, University of South Africa.

5.3.2 Bacterial Strains

The bacterial strain (*Bacillus subtilis* (BD233)) used in this study was obtained from the National Collection of Bacterial Species of the Agricultural Research Council, Plant Protection Research Institute, Roodeplaat, South Africa. Strains previously grown on Luria-Bertani solid media was sub-cultured aseptically in a sterile laminar flow and grown on Luria-Bertani broth at 37°C overnight. Thereafter, pure strains of the bacteria cultures were diluted and the broth adjusted to 0.5 McFarland Standards with sterile distilled water before use for inoculation of cowpea plants.

5.3.3 Plant Growth Material and Cowpea Fertilization

As described in Chapter 4, Sections 4.3.2 and 4.3.3.

5.3.4 Co-Inoculation of Cowpea

As described in Chapter 4, Section 4.3.3, a commercial powered inoculant with *Bradyrhizobium spp* CB756 strains was used for inoculation of cowpea seedlings. In addition, 2 mL of the adjusted (0.5 McFarland standards) *B. subtilis* broth media was applied to roots of each cowpea seedling grown in the glasshouse for the experiment.

5.3.5 Acid Mine Drainage

Acid mine drainage was acquired from an abandoned gold mine in Randfontein in Gauteng Province, South Africa. For the acid mine water (AMD) irrigation treatment, 75% mine water was prepared by mixing 750 mL of AMD with 250 mL tap water and was used to irrigate cowpea seedlings two weeks after planting in the glasshouse. The seedlings were thereafter irrigated with 75% AMD on a weekly basis until termination of the experiment. A previous study showed that 75% AMD considerably enhanced the growth of two potato genotypes although tubers were contaminated with some heavy metals (Nemutanzhela et al., 2017).

5.3.6 Experimental Treatments: *B. subtilis* Inoculation and AMD Irrigation

Prior to planting, seeds were surface-sterilized as described by Pule-Meulenberg et al. (2010). The seeds were divided into batches and each batch was planted in pots filled with acid washed sand. In each pot, three seeds of each cowpea genotype were planted and after seedling establishment were thinned to one plant per pot. The pots containing seedlings with the treatments were placed on top of tables in the glasshouse. In this experiment, the following four treatments were employed; i) Inoculation with *B. subtilis* (BD233) (B+) and irrigation with mine water (75% AMD) (+), ii) Inoculation *B. subtilis* (BD233) inoculation (B+) and no mine water irrigation (-), iii) No *B. subtilis* (BD233) inoculation (B-) and irrigation with mine water (+) and iv) No inoculation with *B. subtilis* (BD233) (B-) and no irrigation with mine water (-) (control).

A factorial (three factors: genotype x *B. subtilis* inoculation x AMD irrigation) experiment was conducted using a randomized complete block design with three replications. The experiment was carried out in an automated glasshouse at the UNISA's Florida science campus.

Data on growth parameters was collected at seven-day intervals. Cowpea nodulation, yield (number of pods, pod weight (fresh and dry) and number of grains per plant and its nutritional quality (elemental composition in shoot and grains) were assessed for each treatment.

5.3.7 Cowpea Growth, Nodulation and Yield Determination

For the first objective, data on growth parameters was collected at seven-day intervals. Cowpea biomass (shoot and root), nodulation (number of nodules per plant), yield (the number of pods, pod weight (fresh and dry)) and the number of grains per plant were determined at 125 DAP.

5.3.7.1 Number of branches, plant height and stem diameter

The number of branches per plant was counted physically every week. The plant height was measured using a measuring tape and the stem diameter using a Vernier calliper at seven-day intervals throughout the experimental period.

5.3.7.2 Plant biomass (shoot and root weights)

In this experiment, six plants for each treatment were randomly harvested 125 days after planting for the determination of the plant's shoot and root fresh and dry weights. A weighing balance was used to determine the shoot and root fresh weights immediately after washing with tap water of all the harvested plant parts (shoot and roots). For dry weight determination, the plant parts were oven-dried at 60°C for 72 h to constant weight and the weight measured using a weighing balance.

5.3.7.3 Nodulation and yield

Four plants were randomly selected and carefully uprooted at 125 DAP, with root system intact to determine nodulation. Harvested roots were washed with water and nodules were carefully detached and their total numbers counted per treatment. The detached nodules were kept frozen at -10°C in a refrigerator. For yield determination for each treatment, the number of pods per plant were counted and the number of grains per pod for each plant also determined.

5.3.8 Measurement of Stomatal Conductance and Chlorophyll Content

The measurement for stomatal conductance was done at 77 DAP and for chlorophyll content at 70 DAP and 84 DAP. For each treatment, a porometer (SC–1Leaf Porometer, Pullman, USA) was used to measure the leaf stomatal conductance (on the abaxial and adaxial sides of a mature leaf). The same fourth leaf from the apex (identified with a marker) of the plant was used to determine both the stomatal conductance and chlorophyll content. A handheld chlorophyll meter (Opti-Sciences model CCM-300, Hudson, USA) was used for measurements. From 70 DAP through to 120 DAP, chlorophyll content was measured every week.

5.3.9 Determination of Mineral Content in Mine Water (75% AMD), Cowpea Shoot and Grain

Three replicate acid mine water (75%) samples were analysed using Inductively Coupled Plasma Mass Spectrometer (ICP-MS) and inductively coupled plasma atomic emission spectroscopy (ICP-AES) to determine the concentration of some elements. For elemental determination in mine water, an aliquot of a well-mixed, homogeneous aqueous sample was accurately measured for sample processing. For total recoverable analysis of an aqueous sample containing undissolved material, analytes were first solubilized by gentle refluxing with nitric and hydrochloric acids. After cooling, the sample was made up to volume, mixed and centrifuged or allowed to settle overnight prior to analysis. For elemental determination in plant organs, three plants per treatment were harvested at 125 DAP and the parts (shoot and grains) separated. Each part was oven-dried (60°C), weighed, and each ground to fine powder (0.85 mm mesh size) prior to analysis for mineral density. To measure the mineral elements (P, Ca, Cu, Zn, Mn, Fe, and Al) in cowpea shoot and grain, 1 g of ground sample was ashed in a porcelain crucible at 500°C overnight. This was followed by dissolving the ash in 5 mL of 6 M HCI (analytical grade) and placing it in an oven at 50°C for 30 min and 35 mL of deionized water was added. The mixture was filtered through Whatman no. 1 filter paper. Mineral element concentration in plant extracts was determined from three replicate samples using inductively coupled plasma-mass spectrometer (ICP-MS) (Ataro et al., 2008). The quality of data collected was checked using standard solutions with certificate of analysis. In place of analyte isotopes to monitor each element, known sample as standard was used after every 10 samples. Sulphur was determined by wet digestion procedure using 65 % nitric acid (high purity grade). In each case, 1 g of milled plant material was digested overnight with 20 mL of 65 % nitric acid in a 250 mL glass beaker. The beaker containing the extract was then placed on a sand bath and gently boiled until approximately 1 mL of the extract was left. After that, 10 mL of 4 M nitric acid (high purity grade) was added and boiled for 10 min. The beaker was removed from the sand bath, cooled, and the extract washed completely in a 100 mL volumetric flask and the extract filtered through Whatman no.2 filter paper. The S in the sample was then determined by direct aspiration on the calibrated simultaneous ICP-MS.

5.3.11 Data Analysis

Data on growth parameters, nodulation, physiological measurements (leaf stomatal

conductance and chlorophyll content) and elemental quality of cowpea were subjected to analysis of variance (ANOVA) using Statistica v. 10, StatSoft (StatSoft Inc., Tulsa, OK, USA). A 3-way ANOVA was performed to compare the interactive effect of genotype x *B. subtilis* inoculation x AMD irrigation on the growth, nodulation, physiology and nutritional quality of cowpea and 1-way ANOVA for comparing mineral nutrients among genotypes and between inoculation and irrigation. All parameters and measurements were tested at p<0.05 significance level and the Duncan multiple range test was used for separation of means among treatment.

5.4 RESULTS AND DISCUSSION

5.4.1 Interactive Effect of *Bacillus subtilis* (BD233) and Acid Mine Water Irrigation on the Growth, Nodulation and Physiology of Cowpea

Climate change has resulted in increases in dry conditions in most regions of South Africa, a water-scarce country. In the advent of a water crisis, it is highly likely that resource-poor farmers will use acid mine water for crop irrigation. Acid mine drainage is noted to contain pollutants such as heavy metals that can affect the growth of crops. In this study, analysis of the mine water (75% AMD) showed relatively low concentrations of some heavy metals (Table 5.1) compared to the South African National guidelines for the irrigation of crops (Department of Water Affairs and Forestry (DAWF), 1996).

		South
	Elemental	African
	composition	Standards
Element	(mg/L)	(mg/L)
Cd	0.0005	0.01–0.05
Li	1.597	>2.5
Se	0.010	0.02–0.05
As	0.002	0.01–2.0
Zn	0.008	1.0–5.0
Ni	0.210	0.20–2.0
Mn	1.350	0.02–10.0
Cr	0.003	0.10–1.0
Pb	0.0002	0.2–2.0

Table 5.1 The concentration of heavy metals in mine water (75% AMD)

However, long-term irrigation of plants with this treatment (low concentrations of heavy metals as was the case in this study) cannot only lead to accumulation of heavy metals in soils but also their uptake by plants (Nemutanzhela et al., 2017). There is a need to bio-remediate AMD for irrigation of crops as a cheaper and environmentally-friendly alternative. The aim of this study was to evaluate the interactive effect of Bacillus subtilis (BD233) inoculation and mine water irrigation on the growth, nodulation, physiology and shoot/grain elemental guality of cowpea. The results show that the interactive effect of *B. subtilis* (BD233) inoculation and mine water irrigation was significant for cowpea's growth, nodulation, physiology and nutritional quality (shoot and grain) of all three genotypes at 125 DAP under glasshouse conditions. Among the genotypes, with exposure to *B. subtilis* (BD233) inoculation and mine water irrigation, Soronko had a significantly higher number of branches (16.33), plant height (42.58 cm) and biomass (fresh and dry weights of shoot (63.25 g & 11.25 g) and root (35.14 g & 4.06 g) respectively) compared to control. The stem girth of Nyira (115.67 mm) was the biggest followed by Soronko (111.00 mm) and Asetenapa had the lowest values of all measured growth parameters (Table 5.2). Although genotypic differences in cowpea's growth under different conditions have been reported, Belane (2011) found a significant interaction effect of genotype and environment on the growth parameters of 30 cowpea genotypes grown in different agro-ecological climatic conditions. However, in this study, inoculation with *B. subtilis* (BD233) had a significant effect on the growth of cowpea. At 125 DAP, plants inoculated with *B. subtilis* (BD233) had a higher number of branches, plant height and biomass (fresh and dry weights of shoot and root) compared to those not inoculated with the bacteria (Table 5.2). In agreement with this finding, Ogugua et al. (2018) reported that *B. subtilis* (BD233) inoculation significantly enhanced seedlings' growth of sweet pepper (36.3%) and lettuce (14.0%) relative to control. The whole plant biomass (dry weight) of sweet pepper increased by 70.7% after soil inoculation with B. subtilis (BD233) compared with control. Similarly, *B. subtilis* strain GB03 improved shoot (35%) and root (42%) length and biomass (133%) in 20 days old Codonopsis pilosula seedlings (Wu et al., 2016). Furthermore, when tomato was inoculated with *B. subtilis* BEB-ISbs, marked increases were reported in the saleable yield, and weight and length of fruit relative to control (Mena-Violante & Olalde-Portugal, 2007). Jeong et al. (2013) reported a 349% increase in plant dry weight after eight weeks, when Brassica juncea inoculated with P-solubilizing *Bacillus* spp. was compared to control. In addition, Shishido et al. (1996) reported enhanced root growth (18%) in two conifer species inoculated with three

strains of *Bacillus polymyxa* (L6-16R, S20-R, and Pw-2R) when compared to control. Interestingly, all three strains stimulated the growth of both conifer species. A study conducted by Ali et al. (2009b) showed enhanced growth in sorghum when inoculated with a strain of *Pseudomonas* sp. under high temperature conditions compared to those not inoculated. Interestingly, when some legume species (common bean, peanut, soybean and pigeon pea) were co-inoculated with rhizobia and several strains and species of *Bacillus*, marked changes in root architecture and increased nodulation were reported (Schwartz et al., 2013). Generally, microorganisms including *Bacillus subtilis* are known to influence plant growth directly (hormone (auxins, cytokinins and gibberellins) secretion (Ali et al., 2009a; Galaviz et al., 2018)) and indirectly (production of stress inhibiting enzymes, siderophore production and P-solubilization) (Schwartz et al., 2013; Meng et al., 2016).

In this study, irrigation with mine water (75% AMD) had a significant effect on the growth of the tested cowpea genotypes. At 125 DAP, the measured growth parameters of plants irrigated with mine water were significantly lower compared to those irrigated with tap water with the exception of the dry root weight (Table 5.2). Mine water is a source of heavy metals (see Table 5.1) and long-term irrigation of crops can subsequently lead to metal toxicity stress in plants (Ali et al., 2013).

Table 5.2: Effects of *B. subtilis* inoculation and acid mine water (75% AMD) irrigation on the growth parameters of cowpea genotypes at 125 DAP in a glasshouse.

	Growth parameters								
Genotype	Branches /plant	Plant height (cm)	Stem diameter (mm)	Shoot FW	Shoot DW	Root FW	Root DW (a)		
Soronko	16.33a	42.58a	111.00b	63.25a	11.25a	35.14a	4.06a		
Nyira	15.33b	28.25b	115.67a	58.94b	8.81b	34.75b	2.13b		
Asetenapa	7.33c	17.33c	60.50c	30.06c	3.56c	19.19c	1.13c		
Inoculation									
B. subtilis	15.61a	31.39a	97.44a	60.92a	9.96a	33.38a	3.67a		
No B. subtilis	10.39b	27.39b	94.00b	40.58b	5.79b	26.00b	1.21b		
<u>Irrigation</u>									
Acid mine water (AMD)	9.89b	24.00b	90.00 <mark>a</mark>	42.46b	6.67b	25.71b	2.17a		
No AMD	16.11a	34.78a	1.44 <mark>b</mark>	59.04a	9.08a	33.67a	2.71a		
F-Statistics									
Genotype (G)	244.44***	283.92***	1698.71***	9.65***	10.22***	6.30***	3.39*		
Inoculation (IN)	205.52***	21.34***	16.27***	9.81***	8.63**	2.97**	6.90**		
Irrigation (IR) G x IN x IR	281.73*** 44.69***	154.31*** 12.86***	178.32*** 2264.49***	6.11*** 15.36***	2.90* 6.61**	3.46** 13.98***	0.31 ^{ns} 1.61*		

Values (M±S.E. (n=3)) followed by similar letters in a column are not significantly different at *p \leq 0.05, **p \leq 0.01, ***p \leq 0.001 and ns = not significant.

Similarly, at 125 DAP, the interactive effect of *B. subtilis* (BD233) inoculation and mine water irrigation was significant on the growth of each cowpea genotype. For example, the highest number of branches per plant was obtained by Asetenapa (21.67) followed by Nyira (21.00) when not inoculated with *B. subtilis* and irrigated mine water compared to the same genotypes exposed to other treatments. In contrast, with or without *B. subtilis* (BD233) inoculation, Asetenapa irrigated with mine water had zero number of branches (0.00) because the seedlings did not survive exposure to mine water (Table 5.3). Accumulation of toxic concentration of some heavy metals in mine water (Table 5.1) could have caused death of seedlings. For example, even at low concentrations, cadmium can cause growth reduction, wilting, chlorosis and cell death in plants (Gallego et al., 2012). However, when Soronko and Nyira were exposed to the same treatment, they showed some degree of tolerance although the mechanism involved may be still unknown.

At 125 DAP, the measured growth parameters such as plant height, stem diameter, shoot fresh (FW) and dry (DW) weight and root fresh (FW) and dry (DW) weight were

affected by the interaction of *B. subtilis* (BD233) inoculation and mine water irrigation. For example, without mine water irrigation, Soronko inoculated with *B. subtilis* (BD233) had the tallest plants (111.0 cm) compared to those not inoculated with the bacteria (6.7 cm). In essence, inoculation with the bacteria enhanced Soronko plant height by over 60% compared to uninoculated plants. Similarly, Asetenapa (108.0 cm) inoculated with B. subtilis (BD233) but not irrigated with mine water had the second tallest plants. However, Asetenapa not inoculated with the bacteria and not irrigated with mine water had a significantly wider stem diameter (128.0 mm) compared to the other treatments (Table 5.3). Biomass determination also showed the interaction effect of *B. subtilis* (BD233) inoculation and mine water irrigation. For example, Soronko inoculated with *B. subtilis* (BD233) and not irrigated with mine water had the highest shoot fresh and dry weights (104.5 g & 18.3 g respectively) and root fresh and dry weights (54.5 g & 8.0 g respectively) compared to other treatments. Similarly, the biomass of Nyira was enhanced when inoculated with B. subtilis (BD233) but not irrigated with mine water compared to plants inoculated with the bacteria and irrigated with mine water (Table 5.3).

			Growth parameters						
Genotype	Inoculation	Irrigation	Branches/ plant	Plant height (cm)	Stem diameter (mm)	Shoot FW (g)	Shoot DW (g)	Root FW (g)	Root DW (g)
Soronko	B+	+ -	18.0ba 17.0c	69.7ab 111.0a	102.0g 117.0c	89.5ab 104.50a	17.0ab 18.3a	41.5ab 54.7a	6.5ab 8.0a
Soronko	B-	+ -	12.3d 18.0bc	103.0a 6.7c	120.0bc 105.0fg	47.3cd 11.8de	9.8bcd 1.0f	30.8bc 13.5cd	1.8bc 1.0c
Nyira	B+	+ -	10.3c 19.0b	106.0a 37.3b	123.0b 108.0ef	44.0cd 90.8ab	8.8cde 14.3abc	27.5bc 46.0ab	2.8bc 3.8abc
Nyira	B-	+ -	11.0dc 21.0a	72.0ab 72.0ab	120.7bc 111.0dc	37.32cd 63.8bc	3.0def 9.3bcd	24.0bc 41.5ab	1.5c 2.0bc
Asetenapa	B+	+ -	0.0g 7.7f	0.0c 108.0a	0.0h 114.0d	0.0e 36.8cd	0.0f 1.5ef	0.0d 30.5bc	0.0c 1.0c
Asetenapa	B-	+ -	0.0g 21.7a	0.0c 74.3ab	0.0h 128.0a	0.0e 83.5ab	0.0f 12.8abc	0.0d 46.2ab	0.0c 3.5abc

Table 5.3: Interactive effects of *B. subtilis* inoculation and acid mine water (75% AMD) irrigation on the growth parameters of cowpea at 125 DAP in a glasshouse.

Values (M±S.E. (n=3)) followed by similar letters in a column are not significantly different at *p \leq 0.05, **p \leq 0.01, ***p \leq 0.001 and ns = not significant. B+ (inoculation with *B. subtilis* (BD233)), B- (not inoculated with *B. subtilis* (BD233)) and + (irrigation with mine water (75% AMD) and – (not irrigated with mine water).

The nodulation and yield of the tested cowpea genotypes varied significantly in response to *B. subtilis* inoculation and irrigation with mine water. Among the genotypes, Soronko had the highest number of nodules per plant (15.75), followed by

Nyira (12.94) and the least was in Asetenapa (3.94) (Table 5.4). Also, the result showed that Soronko had significantly higher number of pods per plant (5.06), pod fresh and dry weights (12.88 g & 3.88 g respectively), number of grains per plant (23.81) and grain fresh weight per plant (3.56 g) compared to Asetenapa that had the lowest values of these measured parameters (Table 5.4).

According to Table 5.4, one-way analysis revealed that inoculation with *B. subtilis* (BD233) significantly affected cowpea nodulation and yield. Plants co-inoculated with *B. subtilis* (BD233) had significantly higher number of nodules compared to those not inoculated with the bacteria. Similarly, the number of grains and grain fresh weight/plant of plants co-inoculated with *B. subtilis* (BD233) were significantly higher compared to the those not co-inoculated with *B. subtilis* (BD233) and the number of pods and pod biomass (fresh and dry weights) per plant were slightly more than those of plant not co-inoculated with *B. subtilis* (BD233) (Table 5.4).

Similarly, irrigation with mine water (75% AMD) had a significant effect on the nodulation and yield of the tested cowpea genotypes. The nodulation (number of nodules/plant) and measured yield parameters (number of pods/plant, pod fresh and dry weights, number of grains and grain fresh weight/plant) of plants irrigated mine water were significantly lower compared to those not irrigated with mine water (Table 5.4).

		Nodulation and yield							
	No. of	No. of	Pods FW	Pods DW	No. of	Grain FW			
<u>Genotype</u>	nodules/plant	pods/plant	(g)	(g)	grains/plant	(g)/plant			
Soronko	15.75±5.9a	5.06±0.75a	12.88±2.12a	3.88±0.2a	23.81±2.5a	3.56±0.7a			
Nyira	12.94±5.1b	4.88±1.28b	10.88±0.12b	2.13±0.2a	23.75±2.3a	3.38±0.8a			
Asetenapa	3.94±2.1c	1.88±0.89c	4.13±0.37c	1.50±0.1a	7.75±6.8b	1.31±0.1b			
Inoculation									
B. subtilis	15.00±4.5a	4.13±0.87a	10.54±2.37a	3.29±0.2a	22.88±2.1a	3.33±0.2a			
No B. subtilis	6.75±2.8b	3.75±0.85a	8.04±0.74a	1.71±0.8a	14.00±1.3b	2.17±0.2b			
Irrigation Acid mine water									
(AMD)	4.67±1.8b	2.25±0.56b	6.46±1.65b	0.96±0.1b	9.21±0.9b	2.04±0.2b			
No AMD	17.08±4.8a	5.63±0.96a	12.13±2.54a	4.04±0.3a	27.67±2.6a	3.46±0.3a			
F-Statistics									
Genotype (G)	2.24*	7.08*	4.89*	2.37 ^{ns}	6.68**	5.79**			
Inoculation (IN)	3.30*	0.23 ^{ns}	1.09 ^{ns}	2.93 ^{ns}	4.73*	3.79*			
Irrigation (IR)	7.43*	18.91***	5.59*	11.13*	20.47***	5.59*			
G x IN x IR	1.85*	5.62**	3.63*	6.37**	5.71**	9.61***			

Table 5.4: Effects of *B. subtilis* inoculation and acid mine water (75% AMD) irrigation on nodulation and yield of cowpea genotypes at 125 DAP in a glasshouse.

Values (M±S.E. (n=3)) followed by similar letters in a column are not significantly different at *p≤0.05, **p≤0.01, ***p≤0.001 and ns = not significant.

Similarly, the interactive effect of *B. subtilis* (BD233) and mine water irrigation was significant on the nodulation and yield of the tested cowpea genotypes. For example, at 125 DAP, without mine water irrigation, Soronko (41.5) followed by Nyira (25.3), had the utmost nodule counts per plant when inoculated with *B. subtilis* compared to the same genotypes exposed to other treatments or their respective controls (Table 5.5). Again, without mine water irrigation, the same genotypes, Soronko and Nyira, when inoculated with *B. subtilis* had the uppermost pod counts per plant (8.0 & 8.5 respectively), pod fresh weight (19.0 g & 21.3 g respectively), pod dry weight (10.3 g & 6.3 g respectively), number of seeds per plant (53.0 & 40.8 respectively) and seeds fresh weight (6.8 g & 5.3 g respectively) compared to the other treatments and their respective controls (Table 5.5). In agreement with this finding, preliminary investigation indicated that the interactive effect of *B. subtilis* (BD233) inoculation and elevated temperature was significant for the germination and plumule length of cowpea (Nevhulaudzi et al., 2017). Taken together, these results are in consonant with other related studies in which the interactive effect of bacteria on growth of plants has been

reported. For example, the interaction of some plant growth promoting bacteria and P fertilizers promoted the production of dry matter in chickpea (Gupta et al., 2009), rice and wheat (Sharma & Prasad, 2003), and sunflower (Joe & Sivakumaar, 2009). Pea plants co-inoculated with *Bacillus* PZ3 was reported to boost the plant's growth, nodulation, leghaemoglobin, chlorophyll and protein content compared to control plants (Wani & Zainab, 2016). Similarly, the interaction of *Bacillus subtilis* and seaweed (kelpak) was reported to be significant for potato's growth and yield (Kanu & Ntushelo, 2015).

Genotype	Inoculation	Irrigation	No. of nodules/ plant	No. of pods/ plant	Pods FW (g)	Pods DW (g)	No of seeds/plant	Seeds FW/pl ant
Soronko	B+	+	18.2ab	6.0ab	16.3ab	2.8bc	24.3bcd	3.5ab
		-	41.5a	8.0a	19.0a	10.3a	53.0a	6.8a
Soronko	В-	+	2.7d	4.8abc	14.3b	2.5bc	11.8cde	3.5ab
		-	0.5d	1.5cd	2.0c	0.1c	6.3de	0.5bc
Nyira	B+	+	3.2d	2.0bcd	5.0c	0.5c	19.3bcde	4.5a
		-	25.3ab	8.5a	21.3a	6.3ab	40.8ab	5.3a
Nyira	В-	+	0.0d	0.5cd	1.5c	0.0c	0.0e	0.0c
-		-	23.0ab	8.5a	15.8b	1.8bc	35.0ab	3.8a
Asetenapa	B+	+	0.0d	0.0d	0.0d	0.0c	0.0e	0.0c
-		-	3.8d	0.3cd	1.8c	0.0c	0.0e	0.0c
Asetenapa	В-	+	0.0d	0.0d	0.0d	0.0c	0.0e	0.0c
		-	12.0c	7.3a	14.8b	6.0ab	31.0bc	5.3a

Table 5.5: Interactive effects of *B. subtilis* inoculation and acid mine water (75% AMD) irrigation on nodulation and yield of cowpea genotypes at 125 DAP in a glasshouse.

Values (M±S.E. (n=3)) followed by similar letters in a column are not significantly different at *p≤0.05, **p<0.01, ***p≤0.001 and ns = not significant. B+ (inoculation with *B. subtilis* (BD233)), B- (not inoculated with *B. subtilis* (BD233)) and + (irrigation with mine water (75% AMD) and – (not irrigated with mine water).

Plants use stomata to regulate transpiration, metabolite flux, and water retention. In this study, the data obtained showed that the interaction of *B. subtilis* (BD233) and mine water (75% AMD) irrigation significantly affected leaf stomatal conductance of cowpea genotypes. For example, when irrigated with tap water, Nyira inoculated with *B. subtilis* (BD233) had the highest leaf stomatal conductance on the adaxial (120.00 \pm 12.1 m²/s mol⁻¹) and abaxial (123.00 \pm 12.8 m²/s mol⁻¹) sides of the leaf compared to when inoculated and irrigated with mine water (adaxial: 106.00 \pm 10.5 m²/s mol⁻¹) and abaxial (123.00 \pm 12.8 m²/s mol⁻¹) sides inoculated with the bacteria and treated with mine water, both the adaxial and abaxial stomatal conductance were reduced compared to Soronko plants inoculated with the bacteria but irrigated with tap water (Table 5.6). Compared with all other treatments, Soronko not inoculated with *B. subtilis* (BD233) and irrigated with mine water had the lowest

stomatal conductance. In general, all three genotypes with or without B. subtilis (BD233) inoculation had lower stomatal conductance when irrigated with mine water compared to when not irrigated AMD. Interestingly, several authors (Du Plessis, 1983; Jovanovic et al., 1998; Pandey & Sharma, 2002; Ali et al., 2013; Özkay et al., 2014; Nemutanzhela et al., 2017) have reported that long-term irrigation with mine water can increase concentrations of metals in soils, which plants can subsequently take-up and bio-accumulate in their organs. In particular, accumulation of high concentration of heavy metals especially at toxic levels can results in changes in plant physiological processes. For instance, high accumulation of Cd in *Brassica napus* leaf drastically reduced water balance, stomatal conductance and CO₂ concentrations (Ali et al., 2013). This report is in line with findings in the current study as long-term irrigation with mine water with heavy metals (Table 5.1) reduced the stomatal conductance of two cowpea genotypes (Soronko and Nyira) regardless of inoculation with or without B. subtilis (BD233) and most probably resulted in death of Asetenapa seedlings (represented as not determined (nd)) (Table 5.6). Furthermore, the results seem to suggest that inoculation with *B. subtilis* (BD233) could have influenced concentration of heavy metals uptake from the rhizosphere soils, as cowpea plants inoculated with the bacteria had higher stomatal conductance compared to those not inoculated. For example, Soronko irrigated with mine water had higher stomatal conductance when inoculated with *B. subtilis* (BD233) (102.00±10.5 (adaxial) and 105.00±10.5 (abaxial)) compared to without inoculation with the bacteria (71.33±7.1 (adaxial) and 102.00±10.5 (abaxial)) (Table 5.6). This finding is in agreement with reports that Bacillus spp. can remediate heavy metals in contaminated soils (Wani et al., 2007; Fatnassi et al., 2014).

			Stomatal conductance (m ² /s mol ⁻¹)				
Genotype	Inoculation	Irrigation	Adaxial	Abaxial			
Soronko	B+	+	102.00±10.5c	105.00±10.5h			
		-	115.00±11.5ab	117.00±11.8c			
Soronko	В-	+	71.33±7.1d	102.00±10.5g			
		-	116.67±15.8ab	120.00±11.5b			
Nyira	B+	+	106.00±10.5bc	108.00±10.8f			
		-	120.00±12.1a	123.00±12.8a			
Nyira	B-	+	107.00±10.5bc	111.00±10.5e			
		-	109.00±10.5bc	125.00±12.5e			
Asetenapa	B+	+	nd	nd			
		-	112.00±11.5b	114.00±11.5d			
Asetenapa	B-	+	nd	nd			
		-	120.00±12.2a	123.33±12.1a			
F-Statistics							
G x IN x IR			49.09***	79.28***			

Table 5.6: Interactive effects of *B. subtilis* inoculation and mine water (75% AMD) irrigation on leaf stomatal conductance (m²/s mol⁻¹) of cowpea grown in a glasshouse at 77 DAP.

Values (M±S.E. (n=3)) followed by similar letters in a column are not significantly different at *p \leq 0.05, **p \leq 0.01, ***p \leq 0.001 and ns = not significant. B+ (inoculation with *B. subtilis* (BD233)), B- (not inoculated with *B. subtilis* (BD233)) and + (irrigation with mine water (75% AMD) and – (not irrigated with mine water) and nd (not determined).

Similarly, the interaction of *B. subtilis* (BD233) inoculation and acid mine water (75%) AMD) irrigation significantly affected leaf chlorophyll content in cowpea. For example, at 70 DAP, with mine water irrigation, Asetanapa not inoculated with B. subtilis (BD233) recorded the highest chlorophyll content $(123.00 \pm 11.8 \text{ mg/m}^2)$ (chlorophyll a) and 121.00±0.58 mg/m²) (chlorophyll b)) compared to when it was inoculated with the bacteria $(107.33\pm10.1 \text{ mg/m}^2)$ (chlorophyll a) and $71.00\pm7.1 \text{ mg/m}^2$) (chlorophyll b)) (Table 5.7). Two weeks later, at 84 DAP, a similar trend was noted. Asetanapa not inoculated with the bacteria but irrigated with mine water recorded the highest chlorophyll content (124.00±11.58 mg/m²) (chlorophyll a) and 124.33±12.33 mg/m²) (chlorophyll b)) compared to when inoculated with *B. subtilis* (BD233) (110.00±10.5 mg/m²) (chlorophyll a) and 80.33±8.1 mg/m²) (chlorophyll b)) (Table 5.7). At both 70 and 84 DAP, with or without *B. subtilis* (BD233) inoculation, irrigation with mine water reduced the chlorophyll contents in all three tested cowpea genotypes. This finding is in consonant with reports by several authors (Pandey & Sharma, 2002; Zengin & Munzuroglu, 2006; Elloumi et al., 2007; John et al., 2009; Nemutanzhela et al., 2017) that long-term exposure of plants to heavy metal contaminated soils can induce stress known to reduce chlorophyll content due to metal toxicity. Furthermore, the interaction of metals could also negatively influence the leaf chlorophyll content. For example, Soltangheisi et al. (2014) reported that at 28 days after transplanting, the interactive effect of Zn and Fe in rhizosphere soil decreased the chlorophyll content of sweet corn.
Table 5.7: Interactive effects of *B. subtilis* inoculation and acid mine water (75% AMD) irrigation on leaf chlorophyll content (mg/m^2) of cowpea grown in a glasshouse.

			Leaf chlorophyll (mg/m²)				
			70 DAP		84 DAP		
Genotype	Inoculation	Irrigation	Chlorophyll a	Chlorophyll b	Chlorophyll a	Chlorophyll b	
Soronko	B+	+	70.67±6.8c	102.00±10.5b	41.33±4.1d	102.00±10.5b	
		-	76.67±6.3c	77.00±7.5c	112.33±10.3ab	114.33±11.3ab	
Soronko	В-	+	43.00±3.8d	72.33±7.1c	73.33±7.1c	105.00±10.5b	
		-	114.00±10.8ab	111.33±11.9b	115.00±11.5ab	117.00±10.5ab	
Nyira	B+	+	105.00±10.5b	72.33±7.1c	105.00±10.5b	108.00±10.8b	
		-	117.00±11.5ab	116.00±11.5b	118.00±11.8ab	120.00±12.5a	
Nyira	В-	+	106.67±10.8b	81.33±7.9c	105.67±11.6b	107.33±10.8b	
		-	120.00±12.5a	108.33±10.3b	121.00±12.0a	116.67±11.8ab	
Asetenapa	B+	+	nd	nd	nd	nd	
		-	107.33±10.1b	71.00±7.1c	110.00±10.5b	80.33±8.1c	
Asetenapa	В-	+	nd	nd	nd	nd	
		-	123.00±11.8a	121.00±0.58a	124.00±11.58a	124.33±12.33a	
F-Statistics							
G x IN x IR			13.64***	8.61**	35.49***	39.33***	
Values (M±S.E. (n=3)) followed by similar letters in a column are not significantly different at *p≤0.05, **p≤0.01,							

*** $p \le 0.001$ and ns = not significant. B+ (inoculation with *B. subtilis* (BD233)), B- (not inoculated with *B. subtilis* (BD233)) and + (irrigation with mine water (75% AMD) and – (not irrigated with mine water) and nd (not determined).

5.4.1 Effect of *Bacillus subtilis* (BD233) and Acid Mine Water Irrigation on the Nutritional Content of Cowpea's Shoot and Grain

One-way analysis of variance revealed significant differences in the nutritional content among shoots of the tested cowpea genotypes in response to *B. subtilis* (BD233) inoculation and mine water irrigation. For example, the highest concentrations of P (0.49%), Ca (5.65%), Fe (271.13 mg/kg), Cu (2.51 mg/kg), Zn (80.54 mg/kg) and Mn (103.38 mg/kg) were found in Nyira shoot compared to Asetenapa, which had the lowest concentrations of these elements (P=0.17±0.05%, Ca=2.48%, Fe=143.61 mg/kg, Cu=1.19 mg/kg, Zn=31.99 mg/kg, and Mn=40.14 mg/kg respectively) (Table 5.8). The concentration of AI was similar in the shoot of both Soronko and Nyira but was markedly higher compared to Asetenapa. Interestingly, inoculation with *B. subtilis* (BD233) significantly affected the concentration of Fe, AI and Mn in cowpea shoot. For example, plants inoculated with *B. subtilis* (BD233) had lower concentration of Fe (185.75 mg/kg) and AI (60.11 mg/kg) compared to those not inoculated with the bacteria (241.17 and 72.66 mg/kg respectively). In contrast, Mn concentration was markedly higher in shoot of cowpea plants inoculated with *B. subtilis* (BD233) (77.10 mg/kg) compared to those not inoculated with the bacteria (67.59 mg/kg) (Table 5.8). Similarly, irrigation with mine water (75% AMD) affected the concentration of some heavy metals in cowpea shoot. For example, plants irrigated with mine water had marked concentrations of Fe (244.64 mg/kg), Zn (64.79 mg/kg) and Al (71.75 mg/kg) compared to those not irrigated with mine water (182.28, 54.18 and 61.02 mg/kg respectively) (Table 5.8). In contrast, Cu concentration was higher in shoot of plants not irrigated with mine water (2.15 mg/kg) compared to those irrigated with mine water (1.91 mg/kg). However, the concentrations of P, Ca and Mn were not different when plants irrigated with mine water was compared to those not exposed to the same treatment (Table 5.8). The high concentrations of these metals (Fe, Zn and Al) found in cowpea shoot could have been derived from the mine water (75% AMD) (Table 5.1). Long-term irrigation of crops with polluted waters such as acid mine water could lead to high concentrations of heavy metals in rhizosphere soils, and subsequently, their uptake and accumulation in plant organs will be enhanced. Compared to control, the lower concentrations of these metals in shoot of cowpea inoculated with *B. subtilis* (BD233) would suggest their metal toxicity tolerance and potential to bio-remediate heavy metal polluted soils as reported by other authors (Khan et al., 2009; Jeong et al., 2013; Singh et al., 2014).

	Mineral composition in shoot (DM)							
			mg/kg					
<u>Genotype</u>	P (%)	Ca (%)	Fe	Cu	Zn	Mn	AI	
Soronko	0.32b	5.29b	225.64b	2.39b	65.92b	73.51b	79.24a	
Nyira	0.49a	5.65a	271.13a	2.51a	80.54a	103.38a	79.17a	
Asetenapa	0.17c	2.48c	143.61c	1.19c	31.99c	40.14c	40.75b	
Inoculation								
B. subtilis	0.32a	4.37a	185.75b	2.07a	60.77a	77.10a	60.11b	
No <i>B. subtilis</i>	0.33a	4.58a	241.17a	1.98a	58.20a	67.59b	72.66a	
Irrigation Mine water (75%								
AMD)	0.34a	4.19a	244.64a	1.91b	64.79a	69.76a	71.75a	
No AMD	0.30a	4.75a	182.28b	2.15a	54.18b	74.93a	61.02b	
F-Statistics								
Genotype (G)	116.01***	31.29***	14.79***	68.93***	95.85***	44.45***	21.10**	
Inoculation (IN)	0.05 ^{ns}	0.24 ^{ns}	8.16***	0.77 ^{ns}	0.76 ^{ns}	3.01 ^{ns}	5.06*	
Irrigation (IR)	5.26***	2.49 ^{ns}	10.32**	5.32*	13.01**	0.89 ^{ns}	3.69*	
G x IN x IR	48.14***	40.11***	17.03***	69.54***	84.55***	38.58***	15.22***	

Table 5.8: Effect of inoculation and irrigation on the mineral composition in shoot (dry matter (DM)) of cowpea genotypes under glasshouse conditions.

Values (M±S.E. (n=3)) followed by similar letters in a column are not significantly different at *p≤0.05, **p≤0.01, ***p≤0.001 and ns = not significant.

In response to *B. subtilis* (BD233) inoculation and mine water irrigation, significant differences were recorded in grain elemental concentrations among the tested cowpea genotypes (Table 5.9). Soronko grain had the highest concentrations of P (0.53%), Ca (0.13%), Fe, (163.89 mg/kg), Cu (3.47 mg/kg), Zn (55.20 mg/kg), Mn (15.04 mg/kg) and AI (2.70 mg/kg) compared to the other genotypes (respectively) (Table 5.9). Inoculation with *B. subtilis* (BD233), as a single factor, significantly affected the accumulation of some heavy metals in grains of cowpea. For example, concentrations of Fe (133.11 mg/kg) and Zn (37.43 mg/kg) were markedly higher in grains of cowpea plants not inoculated with *B. subtilis* (BD233) compared to those inoculated with the bacteria (51.81 and 33.98 mg/kg respectively) (Table 5.9). However, inoculation with *B. subtilis* (BD233) did not significantly impacted grain accumulation of some elements (P, Ca, Cu, Mn and AI) as slight differences were noted compared to uninoculated plants (respectively). In general, plants not inoculated with *B. subtilis* (BD233) accumulated more heavy metals in their grains compared to those inoculated with the bacteria.

		N	Mineral composition in grain (DM)					
			mg/kg					
<u>Genotype</u>	P (%)	Ca (%)	Fe	Cu	Zn	Mn	AI	
Soronko	0.53a	0.13a	163.89a	3.47a	55.20a	15.04a	2.70a	
Nyira	0.42b	0.07b	75.27b	2.29b	40.42b	11.73b	0.45b	
Asetenapa	0.13c	0.03b	38.21c	0.83c	11.49c	2.91c	0.25b	
Inoculation								
B. subtilis	0.36a	0.06a	51.81b	1.93a	33.98b	9.28a	0.43a	
No B. subtilis	0.36a	0.09a	133.11a	2.46a	37.43a	10.51a	1.83a	
<u>Irrigation</u> Mine water (75%								
AMD)	0.45a	0.10a	137.57a	3.05a	46.23a	12.82a	1.88a	
No AMD	0.28b	0.05a	47.34b	1.35b	25.17b	6.97b	0.38a	
F-Statistics								
Genotype (G)	929.21***	5.02*	3.27**	27.47***	327.35***	21.09***	1.51*	
Inoculation (IN)	0.24 ^{ns}	1.12 ^{ns}	3.88*	3.26 ^{ns}	5.91*	0.60 ^{ns}	1.20 ^{ns}	
Irrigation (IR)	464.28***	4.02 ^{ns}	4.78*	33.65***	220.36***	13.81**	1.37 ^{ns}	
G x IN x IR	173.68***	0.04 ^{ns}	0.13 [*]	0.32 ^{ns}	36.33***	2.05 ^{ns}	0.97*	

Table 5.9: Effect of inoculation and irrigation on the mineral composition in grain (dry matter (DM)) of cowpea genotypes under glasshouse conditions.

Values (M±S.E. (n=3)) followed by similar letters in a column are not significantly different at *p≤0.05, **p≤0.01, ***p≤0.001 and ns = not significant.

A comparison of the elemental content of cowpea shoot and grain in response to the interactive effect of *B. subtilis* (BD233) inoculation and mine water irrigation at 125 DAP revealed significant differences between and among genotypes. For example, without mine water irrigation, Soronko accumulated more Fe (359.42 mg/kg) in grain (dry matter) compared to shoot (240.8 mg/kg) without inoculation with *B. subtilis* (BD233) but accumulated more Fe in the shoot (217.0 mg/kg) compared to grain (83.69 mg/kg) when inoculated with the bacteria (Figure 5.1A). In contrast, without mine water irrigation, Nyira accumulated more Fe in shoot compared to grain with or without *B. subtilis* (BD233) inoculation (Figure 5.1A).



Figure 5.1: Comparison of elemental content in cowpea shoot and grain in response to the interactive effect of *B. subtilis* (BD233) inoculation and mine water irrigation. A) Iron, B) Phosphorus, C) Zinc and D) Manganese.

In general, with or with *B. subtilis* (BD233) inoculation and mine water irrigation, the grain of all cowpea genotypes accumulated more P compared to shoot. However, without *B. subtilis* (BD233) inoculation, when Nyira was irrigated with mine water, more P was accumulated in shoot than in the grain and when Asetenapa was irrigated with

mine water and inoculated with *B. subtilis* (BD233), the shoot also accumulated more P compared to the grain (Figure 5.1B). In contrast to P accumulation, more Zn was accumulated in shoot compared to grain for all genotypes regardless of the treatment except when Soronko without inoculation with the bacteria and mine water irrigation was the opposite i.e. more Zn in grain than in shoot (Figure 5.1C). Similarly, the concentration of Mn was more in shoot compared to grain for all tested genotypes regardless of the treatment (Figure 5.1D).

The cowpea crop serves as important sources of cheap protein, carbohydrates and nutrients for human nutrition. In this study, with or without mine water irrigation, inoculation with B. subtilis (BD233) showed concentration of some heavy metals (Fe, Zn, & Mn) in shoot and grains that are comparable to those reported in leaf and grain of most cowpea genotypes by Belane (2011). It will be interesting to test B. subtilis inoculation of cowpea under field conditions because microbes are known to help plant take-up nutrients (Schwartz et al., 2013). In this study, plants not inoculated with B. subtilis (BD233) accumulated more heavy metals when irrigated with mine water compared to those inoculated with the bacteria. The shoot of cowpea plants irrigated with mine water had marked concentrations of Fe, Zn, and Al compared to those not irrigated with mine water (Table 5.8). Among the genotypes, the grains of Soronko had more nutrients (P, Ca, Fe, Cu, Zn, Mn and Al) compared to the other genotypes (respectively) (Table 5.8). In general, inoculation with *B. subtilis* (BD233) affected accumulation of minerals in the shoot and grains of all genotypes and genotypic variations were noted in this study. The noted cowpea genotypic differences in mineral content in their organs are consistent with the findings of other studies for various crop species (Quintana et al., 1996; Grusak & DellaPenna, 1999; Belane & Dakora, 2010; Balane, 2011).

5.5 CONCLUSIONS

In conclusion, this study has showed that inoculation with *B. subtilis* (BD233) and irrigation with mine water (75% AMD) significantly influenced the growth (plant height, stem diameter, plant biomass), nodulation and yield of all the tested cowpea genotypes. However, genotypic variation was noted for all measured parameters in response to the treatments employed in the study. Soronko and Nyira showed some level of tolerance to metal toxicity at 125 DAP compared to the Asetenapa plants that did not survive when irrigated with mine water regardless of inoculation with or without

B. subtilis (BD233). The interaction of *B. subtilis* (BD233) inoculation and mine water irrigation affected the physiology (chlorophyll content and stomatal conductance) of cowpea and genotypic variation was also noted. Furthermore, the study showed that with or without *B. subtilis* (BD233) inoculation there was accumulation of heavy metals (Fe, Zn & Al) in cowpea shoot possibly from the mine water and genotypic variation was also noted. Notably, inoculation with *B. subtilis* (BD233), with or without mine water irrigation, affected the accumulation of heavy metals in cowpea grains. This finding clearly indicate that *B. subtilis* (BD233) has ability to sequester heavy metals in soils polluted with mine water.

6.1 GENERAL DISCUSSION

Water for irrigation of crops is a scarce resource especially in areas that have a history of mining in South Africa. Mining of gold and coal over the years in the country has resulted in the pile-up of huge amounts of mine tailings close to water catchment areas (Jovanovich et al., 1998). AMD generated from mine tailings, and through entrance of water into mine voids, is generally characterized by low pH (2-4), high sulphur concentration (1-2 q/l) and high levels of metals, particularly heavy metals such as Zn. Cd, Cu, Ni, Pb, Mn, Fe, Hg and Cr (Vadapali et al., 2008). When these heavy metals are eventually discharged into water systems (surface and ground waters), they can persist and bio-accumulate in living organisms such as plants, ultimately posing health hazards and environmental risk in the ecosystem (Naicker et al., 2003). Ultimately, when water polluted with acid mine drainage are used for crop irrigation it changes soil properties and processes such as contraction of cation exchange capacity and expansion of anion exchange capacity (Fey, 2001) due to soil acidification. In addition, it can decrease the concentrations of some important basic cations (Ca, Mg, K) and solubility of nutrients that are essential for plant growth and also increases the concentration of some toxic elements (Al and Mn) (Sumner et al., 1990). Following heavy metal build-up in rhizosphere soils, plants will subsequently take up, translocate and accumulate them in their organs (Nemutanzhela et al., 2017). The accumulation of heavy metals in food crop organs have health risk implications if human consume such crops. For example, Khan et. al (2008) found significantly higher concentrations of heavy metal in food crop plants grown in wastewater-irrigated soils compared to those on soils not irrigated with the treatment. The concentrations of heavy metal in the food crop were also reported to exceed the accepted parameters stipulated by the State Environmental Protection Administration (SEPA) in China and the World Health Organization (WHO) (Khan et al. 2008). The accumulation of heavy metals in plant organs (with the associated health risk when consumed by humans) and loss of essential nutrients in soils exposed to acid mine water will eventually translate into poor crop productivity and nutrient quality, thereby posing threats to the security of food and nutrition. Therefore, the contamination of surface waters (streams, rivers, dams) and soils with acid mine water has become a matter of global concern (Mamba et al., 2009). In South Africa, a huge proportion (about 5 million ha) of soils are severely acidic and about 11 million ha are moderately acidic (Venter et al., 2001) possibly due

to pyritic formations in mineral deposits and acid mine acidification. In addition, the percentage of arable land and average annual rainfall for crop irrigation is low (Jovanovich et al., 1998). In the country, agriculture is a major economic activity and availability of clean water is a major concern for both commercial and poor-resource farmers. Agriculture utilises about 70% of water withdrawal hence it is expected that in times of water scarcity farmers will turn into polluted waters as water source especially by poor-resource farmers in rural communities (FAO, 2010). Therefore, the scarcity of good quality water in the country especially for irrigation of crops by resource-poor famers is a major concern especially for rural communities within and around the precinct of mining industries. Recently, the country experienced severe drought in some provinces (Western Cape and Free State) and the looming climate change due to the El Nino phenomenon makes the situation very worrisome. One of the major threats associated with climate change is increase in global average temperatures, which poses additional problems for seed germination and crop development (Ewing, 1981). Temperature is noted to significantly affect seed germination (Rivers & Webber, 1971; White & Monstes, 1993; Craufurd et al., 1996), the rate of seed germination and seedling emergence of most crops including cowpea (Garcia-Huidobro et al., 1982; Covell et al., 1986). This additional threat to household food and nutrition security associated with climate change is worrisome and therefore, there is a need to mitigate climate change scenarios such as short-term elevated temperatures. To this end, agricultural management systems should adopt new strategies or approaches to mitigate climate change and increase food productivity. One suggested strategy is to clean-up contaminated surface waters especially those polluted by acid mine water for crop irrigation. However, the cost is colossal for cleaning AMD contaminated water using conventional methods (reverse osmosis, use of wet lands and lime) beside the negative impact of such technologies on the environment (McCarthy, 2011). There is need to search for cheaper and environmentally-friendly alternatives. Interestingly, modern technologies use plants and microorganisms for decontamination of polluted soils or waters and are therefore considered to be cost-effective and environmentally-friendly (Abou-Shanab et al., 2003; Hooda, 2007; Zhuang et al., 2007). Bacillus spp. are among the microorganisms used to bio-remediate contaminated waters and soils (with heavy metals) (Carrasco et al., 2005; Zhuang et al., 2007) and also as enhancers of seed germination and plant growth (Prathibha & Siddalingeshwara, 2013). Another strategy to mitigate climate change is to cultivate genotypes with traits adaptable to rapid and excessive short-

term changes in climatic conditions. This study evaluated the interactive effects of Bacillus subtilis inoculation and elevated temperature on the germination indices (germination percentage (G%), germination index (GI) and germination rate index (GRI)) and growth of three cowpea genotypes. The results obtained showed that the interaction of *B. subtilis* (BD233) inoculation and elevated temperature significantly (p<0.05) influenced the germination indices and plumule length of cowpea seedlings. For example, Nyira (98.6%) followed by Soronko (84.8%) had the highest germination percentage when their seeds were treated with *B. subtilis* (BD233) and without *B.* subtilis (BD233) respectively, and incubated at an elevated temperature (35°C). With or without *B. subtilis* (BD233), all three genotypes incubated at either 10°C, or 40°C or 45°C had the lowest germination percentages compared to the other temperatures (Table 3.2). In general, Nyira seeds inoculated with *B. subtilis* (BD233) and Soronko seeds without the bacteria obtained the fastest germination rates (lowest GI & GRI values) compared to Asetanapa seeds treated either with or without the bacteria (Table 3.3). Soronko seeds inoculated with B. subtilis (BD233) and incubated at 35°C for seven days had the longest plumule (2.06 cm) compared to the other genotypes under similar conditions (Table 3.4). Clearly, this study has demonstrated the potential of *B. subtilis* (BD233) inoculation to positively influence the germination and growth of cowpea, an important candidate for food/nutrition security in Africa.

In addition, the study showed that elevated temperature (>35°C) can significantly affect the growth and nutritional quality of cowpea at different growth stages (preflowering (40 DAP), flowering (90 DAP) and post-flowering (123 DAP)). For example, at an elevated temperature (35°C), the whole plant biomass (fresh and dry weights), and shoot carbon and crude protein contents at the pre-flowering and flowering stages of cowpea where most affected compared to control (25°C) (Table 4.2 and Figs. 4.3 & 4.4). The results obtained in this study suggest that the pre-flowering and flowering stages of cowpea compared to post-flowering are more susceptible to elevated temperatures (30-35°C). According to these results, under future climate change scenarios, rural farmers cultivating cowpea should adopt cropping management strategies that will protect the flowering stage in the crop's growth cycle especially during periods of changes in climatic conditions (temperature extremes over short periods associated with climate change).

Furthermore, this study has showed that interaction of *B. subtilis* (BD233) inoculation and mine water (75% AMD) irrigation can significantly influence cowpea's growth

(plant height, stem diameter, plant biomass) (Table 5.3), nodulation and yield (Table 5.5). Interestingly, there were genotypic variations noted in response to treatments employed in the study as Soronko and Nyira genotypes showed some level of tolerance to metal toxicity at 125 DAP compared to Asetenapa that did not survive when irrigated with mine water regardless of inoculation with or without B. subtilis (BD233) (Tables 5.3 & 5.5). The interaction of B. subtilis (BD233) inoculation and mine water irrigation similarly affected the physiology (chlorophyll content and stomatal conductance) of cowpea (Tables 5.6 & 5.7). Additionally, the study showed that the interaction of *B. subtilis* (BD233) inoculation and mine water (75% AMD) irrigation can influence the nutritional content of cowpea. For example, with or without *B. subtilis* (BD233) inoculation, irrigation with mine water caused accumulation of heavy metals (Fe, Zn & Al) in cowpea shoot and inoculation with *B. subtilis* (BD233), with or without mine irrigation, affected the accumulation of heavy metals in cowpea grains. These results are in agreement those reported by in a recent study by Israr et al. (2016). Their results showed that the interactive effect of phosphorus fertilization and inoculation with a plant growth promoting rhizobacteria (Pseudomonas putida) was significant for plant nutrient uptake (N, P, and K) that resulted in growth enhancement in chickpea (Israr et al., 2016). In this study, the results showed genotypic variations in response to the interactive effect of *B. subtilis* (BD233) inoculation and mine water (75% AMD) irrigation, which seems to suggest that farmers should consider cultivating cowpea genotypes with metal tolerance when using acid mine water for irrigating their crops under water crisis situations associated future climate change. Additionally, the effect of *B. subtilis* inoculation should inform their choice of the cowpea genotypes.

In conclusion, according to the results obtained in this study, the interactive effect of *B. subtilis* inoculation and mine water irrigation can influence the growth, nodulation, yield, physiology and nutritional quality of cowpea. Furthermore, inoculation with *B. subtilis* has potential to enhance seed germination, growth and nutritional quality of cowpea even when irrigated with acid mine water under future climate change scenarios characterised with short-term extreme temperatures. Taken together, the results in this study suggest that *B. subtilis* (BD233) has the potential to serve as an effective heavy metal resistant bacteria especially in soils polluted with mine water.

6.2 RECOMMENDATIONS

- Soronko genotype seeds inoculated with *B. subtilis* (BD233) and incubated at 35°C was found to be better than the other genotypes with respect to germination rates and growth. However, more research is required to compare it with other elite genotypes and determine if the superior seed quality (vigour) of Soronko would interpret to higher yields under field conditions.
- Additionally, the interactive effect of *B. subtilis* (BD233) inoculation and mine water (75% AMD) irrigation on cowpea's growth, nodulation, yield and nutritional quality should be evaluated under field conditions.
- There is need to decipher the mechanisms involved in the metal resistance of any *Bacillus* spp. in soils polluted acid mine water.
- The study findings may be of significance to resource-poor farmers interested in cowpea cropping under future climate change scenarios.

7 REFERENCES

Abou-Shanab, R.A., Angle, J.S., Delorme, T.A., Chaney, R.L., van Berkum, P., Moawad, H., Ghanem, K. and Ghozlan, H.A. 2003. Rhizobacterial effects on nickel extraction from soil and uptake by *Alyssum murale*. *New Phytologist*. 158(1): 219-224.

Acevedo, E., Silva, P. and Silva, H. 2002. Wheat growth and physiology. In: Curtis, B., Rajaram, S., Gomez-Macpherson, H. (Eds.), Bread Wheat. Food and Agriculture Organization of the United Nations, Rome.

Adams, R.M. 1989. Global Climate Change and Agriculture: An Economic Perspective. *American Journal of Agricultural Economics*. 7(5): 1272-1279.

Adesemoye, A.O., Obini, M. and Ugoji, E.O. 2008. Comparison of plant growth promotion with *Pseudomonas aerugenosa* and *Bacillus subtilis* in three vegetables. *Brazilian Journal of Microbiology*. 39(3): 423-426.

Adetumbi, J.A., Teixeira da Silva, J.A. and Alamu, O. 2011. Cowpea (*Vigna unguiculata* L. Walp) seed germination indices and yield as affected by length of storage. *Seed Science and Biotechnology*. 5(1): 11-14.

Adjei-Nsiah, S., Kuyper, T.W., Leeuwis, C., Abekoe, M.K., Cobbinah, J., Sakyi-Dawson, O. and Giller, K.E. 2008. Farmers' agronomic and social evaluation of productivity, yield and N2-fixation in different cowpea varieties and their subsequent residual N effects on succeeding maize crop. *Nutrition Cycling and Agroecosystems*. 80(3): 199-209.

Agarwal, S.K. 1998. APH Publishing Corporation, New Delhi, India. *Environmental Biotechnology* (1st Ed). p.267-289.

Ahenkora, K., Adu-Dapaah, H.K. and Agyemang, A. 1998. Selected nutritional components and sensory attributes of cowpea (*Vigna unguiculata* L. Walp) leaves. *Plant Foods and Human Nutrition*. 52(3): 221-229.

Ahmad Wani, P., Khan, M. S. and Zaidi, A. 2007. Impact of zinc-tolerant plant growthpromoting rhizobacteria on lentil grown in zinc-amended soil. *Agronomy for Sustainable Development*. 28(3): 449–455.

Akcil, A. and Koldas, S. 2006. Acid Mine Drainage (AMD): causes, treatment and case studies. *Journal of Cleaner Production*. 14(12-13): 1139-1145.

Ali, B., Sabri, A.N., Ljung, K. and Hasnain, S. 2009a. Auxin production by plant associated bacteria: impact on endogenous IAA content and growth of *Triticum aestivum* L'. *Letter of Applied Microbiology*. 48(5): 542–547.

Ali, B., Wang, B., Ali, S., Ghani, M., Hayat, M., Yang, C., Xu, L., and Zhou, W., 2013. 5-Aminolevulinic acid ameliorates the growth, photosynthetic gas exchange capacity, and ultrastructural changes under cadmium stress in *Brassica napus* L. *Journal of Plant Growth Regulation*. 32(3): 604-614.

Ali, S. Z., Sadhya, V., Grover, M., Kishore, N., Rao, L. V. and Venkateswarlu. B. 2009b. *Pseudomonas* sp. strain AKM-P6 enhances tolerance of sorghum seedlings to elevated temperatures. *Biology and Fertility of Soils*. 46(1): 45-55.

Alloway, B.J. and Ayres, D.C. 1993. *Chemical Principles of Environmental Pollution. Blackie Academic and Professional*. An imprint of Chapman and Hall, Oxford, UK. p. 291.

Asante, I.K., Adu-Dapaah, H. and Addison, P. 2006. Variation in contents of crude protein and mineral elements in 32 cowpea (*Vigna unguiculata* (L.) Walp) accessions in Ghana. *Ghana Journal of Agricultural Science*. 39(1):83-86.

Ataro, A., McCrindle, R.I., Botha, B.M., McCrindle, C.M.E. and Ndibewu, P.P. 2008. Quantification of trace elements in raw cow's milk by inductively coupled plasma mass spectrometry (ICP-MS). *Food Chemistry*. 111: 243-248.

Auld, T. D. and O'Connell, M. A. 1991. Predicting patterns of postfire germination in 35 eastern Australian Fabaceae. Australian. *Journal of Ecology*. 16: 53–70.

Awika, J.M. and Duodu, K.G. 2017. Bioactive polyphenols and peptides in cowpea (*Vigna unguiculata*) and their health promoting properties: A review. *Journal of Functional Foods*. 38: 686-697.

Ayan, I., Mut, H., Basaran, U., Acar, A., and Asci, O.O. 2012. Forage Potential of Cowpea (*Vigna unguiculata* L. WALP). *Turkish Journal of Field Crops.* 17(2): 135-138.

Aydinalp, C. and Cresser, M.S. 2008. The Effects of Global Climate Change on Agriculture. *American-Eurasian Journal of Agriculture and Environmental Science*. 3(5): 672-676.

Ayisi, K.K., Nkgapele, R.J. and Dakora, F.D. 2000. Nodule formation and function in six varieties of cowpea (*Vigna unguiculata* L. Walp) grown in a nitrogen–rich field soil in South Africa. *Symbiosis*. 28: 17-31.

Aziz, O., Inam, A., and Samiullah Siddiqi, A.H. 1996. Long-term effects of irrigation with petrochemical industry wastewater. *Journal of Environmental Science and Health*. Part A 31: 2595–2620.

Bado, B.V., Bationo, A. and Cescas, M.P. 2006. Assessment of cowpea and groundnut contributions to soil fertility and succeeding sorghum yields in Guinea savannah zone of Burkina Faso (West Africa). *Biology Fertility and Soils*. 43: 171-176.

Bai, Y., Zhou-Xiao, M. and Smith, D. L. 2003. Enhanced soybean plant growth resulting from coinoculation of *Bacillus* strains with *Bradyrhizobium japonicum*. *Crop Science*. 43: 1774–1781.

Bailly, C. 2004. Active oxygen species and antioxidants in seed biology. *Seed Science Research*. 14(2): 93-107.

Barlow, K.M., Christy, B.P., O'Leary, G.J., Riffkin, P.A. and Nuttall J.G. 2015. Simulating the impact of extreme heat and frost events on wheat crop production: A review. *Field Crops Research*. 171: 109–119.

Barman, S.C., Sahu, R.K., Bhargava, S.K. and Chatterjee, C. 2000. Distribution of heavy metals in wheat, mustard and weed grains irrigated with industrial effluents. *Bulletin of Environment Contamination and Toxicology*. 64:489-496.

Belane, A.K. 2011. Evaluating cowpea genotypes for enhanced N₂ fixation and photosynthetic activity, increased grain yield, and density of Dietarily-important mineral elements. PhD thesis, Tshwane University of Technology. Pretoria.

Belane, A.K. and Dakora, F.D. 2010. Symbiotic N₂ fixation in 30 field-grown cowpea (*Vigna unguiculata* L. Walp.) genotypes in the Upper West Region of Ghana measured using ¹⁵N natural abundance. *Biology Fertility and Soils*. 46: 191-198.

Belane, A.K. and Dakora, F. 2011. Levels of nutritionally-important trace elements and macronutrients in edible leaves and grain of 27 nodulated cowpea (Vigna unguiculata L. Walp.) genotypes grown in the Upper West Region of Ghana. *Food Chemistry*, *125*(1): 99 105.

Bhattacharyya, P.N. and Jha, D.K. 2012. Plant growth-promoting rhizobacteria (PGPR): emergence in agriculture. *Journal World of Microbiology and Biotechnology*. 28: 1327-1350.

Bowen, G.D. and Rovira, A.D. 1999. The rhizosphere and its management to improve plant growth. *Advances in Agronomy*. 66: 1-102.

Bradi, H. 2002. *Heavy Metals in the Environment: Origin, Interaction and Remediation* Volume 6. London: Academic Press.

Bressani, R. 1985. Nutritive value of cowpea. In: Singh, S.R. and Rachie, K.O. (Eds.), *Cowpea Research, Production and Utilization*. UK: Wiley, Winchester, p. 353–359.

Broughton, W.J. and Dilworth, M.J. 1970. Control of leghaemoglobin synthesis in Snake beans. *Biochemistry Journal.* 125: 1075-1080.

Burger, J. 2008. Assessment and management of risk to wildlife from cadmium. *Science of the Total Environment.* 389: 37–45.

Burr, T.J., Schroth, M.N and Suslow, T. 1978. Increased potato yields by treatment of seed pieces with specific strains of *Pseudomonas fluorescens* and *P. putida*. *Phytopathology*. 68: 1377-1383.

Bustamante, G.P. and Ciampi, L.P. 1989. Biological control of bacterial wilt of potato induced by *Pseudomonas solanacearum* E. F. Smith. *Revista de Microbiologia*. 20(1):18–26.

Carrasco, J.A., Armario, P., Pajuelo, E., Burgos, A., Caviedes, M.A., Lopez, R. et. al. 2005. Isolation and characterisation of symbiotically effective *Rhizobium* resistant to arsenic and heavy metals after the toxic spill at the Aznalcollar pyrite mine. *Soil Biology and Biochemistry*. 37: 1131–1140.

Caskey, R.J., Singh, B.B. and Oyewole, B. 2001. Contribution of early season cowpea to late season maize in the savanna zone of West Africa. *Biology Agriculture and Horticulture*. 18: 303-315.

Chandler, V.L., Radicell, J.P., Robbins, T.P., Chen, J. and Turks, D. 1989. Two regulatory genes of the maize anthocyanin pathway are homologous: Isolation of B utilizing R genomic sequences. *Plant Cell*. 1(12): 1175-1183.

Chen, F., Dong, J., Wang, F., Wu F. B., Zhang, G. P., Li, G. M., Chen, Z. F., Chen, J. X. and Wei, K. 2007. Identification of barley genotypes with low grain Cd accumulation and its interaction with four microelements. *Chemosphere.* 67: 2082–2088.

Chipo, M., Loveness, M., Stenly, M. and Revai, M. 2011. Assessment of heavy metal accumulation in wastewater irrigated soil and uptake by maize plants (*Zea Mays* L) at Firle Farm in Harare. *Journal of Sustainable Development*. 4: 132-137.

Coetzee, H., Hobbs, P. J., Burgess, J.E., Thomas, A. and Keet, M. (Eds.). (2010). Mine water management in the Witwatersrand Gold Fields with special emphasis on acid mine drainage. Report to the Inter-Ministerial Committee on Acid Mine Drainage. Pretoria: *Department of Water Affairs and Forestry*. Available from <u>http://www.dwaf.gov.za/Documents/ACIDReport.pdf</u>. (Accessed 24 February 2012).

Commuri, P.D. and Jones, R.D. 2001. High temperatures during endosperm cell division in maize: a genotypic comparison under in vitro and field conditions. *Crop Science*. **41**: 1122–1130.

Compant, S., Van Der Heijden, M.G. and Sessitsch, A. 2010. Climate change effects on beneficial plant-microorganism interactions. *FEMS Microbiology Ecology*. 73: 197-214.

Covell, S., Ellis, R.H., Roberts, E.H. and Summerfield, R.J. 1986. The influence of temperature on seed germination rate in grain legumes. I. A comparison of chick pea, lentil, soyabean and cowpea at constant temperatures. *Journal of Experimental Botany*. *3(5)7:* 705-715.

Craufurd, P.Q., Ellis, R.H., Summerfield, R.J. and Menin, L. 1996. Development in cowpea (*Vigna unguiculata*). I. The influence of temperature on seed germination and seedling emergence. *Experimental Agriculture*. 32(1): 1-12.

Dakora, F.D. and Keya, S.O. 1997. Contribution of Legume Nitrogen Fixation to Sustainable Agriculture in Sub –Saharan Africa. *Soil Biology and Biochemistry*. 29: 809-817.

Dakora, F.D., Aboyinga, R.A., Yahaya, M. and Apaseku, J. 1987. Assessment of N₂ fixation in groundnut (*Arachis hypogaea* L.) and cowpea (*Vigna unguiculata* L. Walp) and their relative N contribution to succeeding maize crop in Northern Ghana. *MIRCEN Journal*. 3: 389-399

Dary, M., Chamber, P.M.A., Palomares, A.J. and Pajuelo, E. 2010. "In situ" phytostabilisation of heavy metal polluted soils using *Lupinus luteus* inoculated with metal resistant plant-growth promoting *rhizobacteria*. *Journal of Hazardous Material*. 177: 323-330.

Department of Water Affairs and Forestry (DWAF). 1996. South African Water Quality Guidelines (2nd Edition), Volume 4. Agricultural Water Use: Irrigation. p. 39-170.

Dougherty, T.C. and Hall, A.W. 1995. *Environmental impact assessment of irrigation and drainage projects*. FAO Irrigation and Drainage Paper 53. ISBN 92-5-103731-0. [Online] Available: <u>http://www.fao.org/docrep/v8350e/v8350e00.htm</u>. (Accessed 30 March 2018).

Du Plessis, H.M. 1983. Using lime treated acid mine water for irrigation. *Water Science and Technology.* 15: 145-154.

Dupuis, L. and Dumas, C. 1990. Influence of temperature stress on in vitro fertilization and heat shock protein synthesis in maize (*Zea mays L.*) reproductive systems. *Plant Physiology*. 94: 665-670.

Ehlers, J.D. and Hall, A.E. 1997. Cowpea (*Vigna unguiculata* L. Walp). *Field Crops Research*: 53: 187-204.

Eitzinger, J., Orlandini, S., Stefanski, R. and Naylor, R. E. L. 2010. Climate change and agriculture: introductory editorial. *Journal of Agricultural Science*. 148: 499–500.

Eitzinger, J., Thaler, S., Schmid, E., Strauss, F., Ferrise, R., Moriondo, M., Bindi, M., Palosuo, T., Rötter, R., Kersebaum, K., Olesen, J., Patil, R., Saylan, L., Caldag, B. and Caylak, O. 2013. Sensitivities of crop models to extreme weather conditions during flowering period demonstrated for maize and winter wheat in Austria. *Journal of Agricultural Science*. 151: 813–835.

Elloumi, N., Ben, F., Rhouma, A., Ben, B., Mezghani, I. and Boukhris, M. 2007. Cadmium induced growth inhibition and alteration of biochemical parameters in almond seedlings grown in solution culture. *Acta Physiogiae Plantarum*. 29: 57–62.

Ewing, E.E. 1981. Heat stress and the tuberisation stimulus. *American Potato Journal*. 58(1): 31-49.

FAO. 2002. The State of Food Insecurity in the World 2001, Rome: FAO.

FAO. 2010. COWPEA: Post-Harvest Operations. Post-harvest Compendium. Accessed 23 January 2019. http://www.fao.org/3/a-au994e.pdf.

Fatnassi, I.C., Chiboub, M., Jebara, M. and Jebara, S.H. 2014. Bacteria associated with different legume species grown in heavy-metal contaminated soils. *International Journal of Agricultural Policy and Research*. 2(12): 460-467.

Felici, C., Vettori, L., Giraldi, E., Forino, L.M.C., Toffanin, A., Tagliasacchi, A.M. and Macro Nuti, M. 2008. Single and coinoculation of *Bacillus substilis* and *Azospirillum brasilense* on *Lycoperscion esculentum*: Effects on plant growth and rhizosphere microbial community. *Applied Soil Ecology.* 10: 260-270.

Fey, M.V. 2001. Effects of excessive soil acidity on the natural resources, agriculture and forestry. In: Plant Soil Interactions at low pH: Proceedings of the 5th International symposium (Keynote submitted for publication).

Freire-Filho, F.R. 2011. Feijão-caupi no Brasil: produção, melhoramento genético, avanços e desafios. Teresina: Embrapa Meio-Norte.

Galaviz, C., Lopez, B.R., de-Bashan, L.E., Hirsch, A., Maymon, M. and Bashan, Y. 2018. Root growth improvement of mesquite seedlings and bacterial rhizosphere and soil community changes are induced by inoculation with plant growth-promoting bacteria and promote restoration of eroded desert soil. *Land Degradation and Development*. 1-4.

Gallego, S.M., Pena, L.B., Barcia, R.A., Azpilicueta, C.E., Iannone, M.F., Rosales, E.P., et al. 2012. Unravelling cadmium toxicity and tolerance in plants: insight into regulatory mechanisms. *Environmental and Experimental Botany*. 83: 33–46.

Garcia-Huidobro, J., Monteith, J. L. and Squire, G. R. 1982. Time, temperature and germination of pearl millet *(Pennisetum typhoides* S. &H.). I. Constant temperature. *Journal of Experimental Botany.* 33(2): 288-296.

Giri, A., Heckathorn, S., Mishra, S. and Krause, C. 2017. Heat Stress Decreases Levels of Nutrient-Uptake and -Assimilation Proteins in Tomato Roots. *Plants*. 6: 1-15. doi:10.3390/plants6010006.

Gonçalves, A., Goufo, P., Barros, A., Domínguez-Perles, R., Trindade, H., Rosa, E.A.S., Ferreira, L. and Rodrigues, M. 2016. Cowpea (*Vigna unguiculata* L. Walp), a renewed multipurpose crop for a more sustainable agri-food system: nutritional

advantages and constraints. *Journal of the Science of Food and Agriculture*. 96: 2941–2951.

Grusak, M.A. and Dellapenna, D. 1999. Improving the nutrient composition of plants to enhance human nutrition and health. *Annual Review of Plant Physiology and Plant Molecular Biology*, 50: 133-161.

Gupta, S.C., Singh, R.P. and Ravinder, V. 2009. Response of chickpea to P levels from different sources with various PSB species. *Legume Research*. 32: 5359–5370

Han, F., Chen, H., Li, X.J., Yang, M.F., Liu, G.S. and Shen, S.H. 2009. A comparative proteomic analysis of rice seedlings under various high-temperature stress. *Biochimica et Biophysica Acta*. 1794: 1625-1634.

Hancock, R.D., Morris, W.L., Ducreux, L.J.M., Morris, J.A., Usma, M., Verrall, S.R., Fuller, J., Simpson, C.G. and Zhang, R. 2014. Physiological, biochemical and molecular responses of the potato (*Solanum tuberosum* L.) plant to moderately elevated temperature. *Plant, Cell and Environment*. 37(2): 439–450.

Hao, H.P., Jiang, C.D., Zhang, S.R., Tang, Y.D. and Shi, L. 2012. Enhanced thermaltolerance of photosystem II by elevating root zone temperature in Prunus mira Koehne seedlings. *Plant and Soil*. 353: 367–378.

Harrison, P.A., Berry, P.M. and Paterson, J.S. 2010. Climate change adaptation and mitigation: synergies, antagonisms and trade-offs for biodiversity and ecosystems services. BOU Conference Proceedings – Climate Changeand Birds. [online]. Available from: http://www.bou.org.uk/bouproc-net/ccb/harrison-etal.pdf. (Accessed 4 April 2018).

Hatfield, J.L. and Prueger, J.H. 2015. Temperature extremes: Effect on plant growth and development. *Weather and Climate Extremes*. 10: 4–10.

Hatfield, J.L., Boote, K.J., Fay, P., Hahn, L., Izaurralde, R.C., Kimball, B.A., Mader, T., Morgan, J., Ort, D., Polley, W., Thomson, A. and Wolfe, D. 2008. Agriculture In: The Effects of Climate Change on Agriculture, Land Resources, Water Resources, and Biodiversity in the United States.

Heckathorn, S.A., Giri, A., Mishra, S. and Bista, D. 2014. Heat stress and roots. In Climate Change and Plant Abiotic Stress Tolerance; Tuteja, N., Gill, S., Eds.; Wiley Blackwell: Weinheim, Germany. p. 109–136.

Hijmans, R.J. 2003. The effect of climate change on global potato production. *American Journal of Potato Research*. 80 (4): 271-279.

Holton., T.A. and Cornish, E.C. 1995. Genetics and biochemistry of anthocyanin biosynthesis. *Plant Cell*. 7: 1071-1083.

Hooda, V. 2007. Phytoremediation of toxic metals from soil and waste water. *Journal* of *Environmental Biology*. 28: 367-376.

Houxuan, Z., Yan, S. and Yi, Z. 1993. 'Estimating the impact of climate change on crop yield in Huang-Huai-Hai Plain by SCCM'. In: Yi Z., Peiyuan, Z., Houxuan, Z. and Zhenyao L. (eds) Climate *Change and its Impact.* Meteorological Press, Beijing, Chinese.

Howie, W. J. and Echandi, E. 1983. Rhizobacteria: influence of cultivar and soil type on plant growth and yield of potato. *Soil Biology and Biochemistry*. 15: 127-132.

Huang, B., Rachmilevitch, S. and Xu, J. 2012. Root carbon and protein metabolism associated with heat tolerance. *Journal of Experimental Botany*. 63: 3455–3465.

Hu, P. J., Qiu, R. L., Senthilkumar, P., Jiang, D., Chen,Z. W., Tang, Y. T. and Liu, F. J. 2009. Tolerance, accumulation and distribution of zinc and cadmium in hyper accumulator, *Potentilla griffithii*. *Environmental and Experimental Botany.* 66: 317–325.

Intergovernmental Panel on Climate Change (IPCC). 2007. Climate Change: Impacts, Adaptation and Vulnerability, Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change (Cambridge University Press, Cambridge, UK), in press. Available from: <u>http://www.ipcc.ch.ipccreports/ar4-syr.htm</u>. (Accessed 9 April 2018).

Israr, D., Mustafa, G., Khan, K.S., Shahzad, M., Ahmad, N. and Masood, S. 2016. Interactive effects of phosphorus and *Pseudomonas putida* on chickpea (*Cicer arietinum* L.) growth, nutrient uptake, antioxidant enzymes and organic acids exudation. *Plant Physiology and Biochemistry*. 108:304-312.

ISTA. 2006. *International Rules for Seed Testing*. International Seed Testing Association, Zurich. p 333.

Jacobsen, B. 2002. Biological control of potato pathogens. *Biology Control Crop Diseases*. 7: 179-189.

Jasper, K.I. and Norman, N.P. 1983. Nutrient contents of raw and cooked cowpea leaves. *Journal Food Science*. 48: 1252-1254.

Joe, M.M. and Sivakumaar, P.K. 2009. Long term survivability of *Azospirillum* coaggregates: Bioinoculation effect on the growth and yield of sunflower. *Agricultura*, 6: 71–77.

Jeong, S., Moon, H.S., Shin, D. and Nam, K. 2013. Survival of introduced phosphatesolubilizing bacteria (PSB) and their impact on microbial community structure during the phytooextraction of Cd-contaminated soil. *Journal of Hazardous Material*. 263(2): 441-449.

John, R., Ahmad, P., Gadgil, K. and Sharma, S. 2009. Heavy metal toxicity: effect on plant growth, biochemical parameters and metal accumulation by Brassica juncea L. *International Journal of Plant Production.* 3: 65–76.

Jovanovic, N.Z., Barnard, R.O., Rethman, N.F.G. and Annandale, J.G. 1998. Crops can be irrigated with lime-treated acid mine drainage. *Water South Africa*: 24: 113–122.

Kadam, S.S., Salunkhe, D.K. and Luh, B.S. 1984. Winged bean in human nutrition. *Critical Reviews in Food Science and Nutrition.* 21(1): 1-40.

Kaiser, H.M. 1991. Climate Change and Agriculture.

Kanu S.A. and Ntushelo, K. 2015. Interactive effects of *Bacillus subtilis* and seaweed (kelpak) on growth, secondary metabolites and yield of potato (*Solanum tuberusom* L.) under glasshouse conditions. Second Conference on Ecology of Soil Microorganism, Microbes as Important Drivers of Soil processes, Prague, Czech Republic. Book of Abstract p153.

Kaplan F., Kopka J., Haskell D.W., Zhao W., Schiller K.C., Gatzke N., Sung D.Y. and Guy C.L. 2004. Exploring the temperature-stress metabolome of Arabidopsis. Plant Physiology. 136: 4159-4168.

Karl, T.R., Diaz, H. and Bamett, T. 1990. Climate Variations of the Past Century and the Greenhouse Effect (a report based on the Fkst Climate Trends Workshop). Rockville, MD: National Climate Program Office/NOAA.

Khan, M.S., Zaidi, A., Wani. P.A. and Oves, M. 2009. Role of plant growth rhizobacteria of contaminated soils. Environment Chemistry Letters. 7:1-19.

Khan, S., Cao, Q., Chen, B., Huang, Y.M. and Zhu, Y.G. 2008. Health risks of heavy metals in contaminated soils and food crops irrigated with wastewater in Beijing, China. *Environmental Pollution.* 152: 686-692.

Kim, J.O., Lee, Y.W. and Chung, J. 2013. The role of organic acids in the mobilization of heavy metals from soil. *KSCE Journal of Civil Engineering.* 17: 1596-1602.

Klopper, J.W., Leong, J., Teintze, M., Schroth, M.N. 1980. *Pseudomonas* siderophores: A mechanism explaining disease suppressive soils. *Curro Microbial*. 4:317-320.

Kloepper, J.W., Ryu, C-M. and Zhang, S. 2004. Induced systemic resistance and promotion of plant growth by *Bacillus* spp. *Phytopathology*. 94: 1259–1266.

Kondache, K., Koffi, D., Kayode, J. A., Awokola, S. O. and Adebola, A. 2018. Impact of climate variability on crop yields in Southern Togo. *Agricultural Development and Sustainable environment*, Nigeria.

Kritzinger, Q. 2004. Antimicrobial activity and fumonisins associated with cowpea (*Vigna unguiculata*). PhD thesis, University of Pretoria, Pretoria. South Africa.

Kurukulasuriya, P., Mendelsohn, R., Hassan, R., Benhin, J., Deressa, T., Diop, M., Eid, H.M., Fosu, K.Y., Gbetibouo, G., Jain, S., Mahamadou, A., Mano, R., Kabubo-Mariara, J., El-Marsafawy, S., Molua, E., Ouda, S., Ouedraogo, M., Se´ne, I., Maddison, D., Seo, S.N. and Dinar, A. 2011. Will African Agriculture Survive Climate Change? *The World Bank Economic Review*. 20(3):367–388.

Lafta, A. M. and Lorenzen, J. H. 1995. Effect of high temperature on plant growth and carbohydrate metabolism in potato. *Journal of Plant Physiology.* 109: 637-643.

Langyintuo, A.S., Lowenberg-Deboer, J., Faye, M., Lambert, D., Ibro, G., Moussa, B., Kergna, A., Kushwaha, S., Musa, S. and Ntoukam, G. 2003. "Cowpea supply and demand in West and Central Africa". *Field Crops Research*. 82: 215–231. doi:10.1016/s0378-4290(03)00039-x.

Levy, D. and Veilleux, R.E. 2007. Adaptation of potato to high temperatures and salinity: A review. *American Journal of Potato Research*. 84: 487-506.

Mamba, B.B., Dlamini, N.P., Nyembe, D.W. and Mulaba-Bafubiandi, A.F. 2009. Metal adsorption capabilities of clinoptilolite and selected strains of bacteria from mine water. *Physics and Chemistry of the Earth.* 34: 830–840.

Manjula, K. and Podile, A.R. 2001. Chitin-supplemented formulations improve biocontrol and plant growth promoting efficiency of *Bacillus subtilis* AF 1. *Canadian Journal of Microbiology*. 47(7): 618-625.

Marchel, M., Hajduk, E. and Kaniuczak, J. 2015. *Arabidopsis thaliana* reaction in response to cadmium and *Trichoderma* fungi additive into the soil. *American Journal of Experimental Agriculture.* 10(3): 1-9.

McCarthy, T.S. 2011. The impact of acid mine drainage in South Africa. South African Journal of Science.107(5/6): 712-719. doi:10.4102/sajs.v107i5/6.712.

McIlveen, R. 2010. Cutting the Cost of Cutting Carbon, in Greener, Cheaper, S. Less, ed., Policy Exchange, London.

Mena-Violante, H.G., and Olalde-Portugal, V. 2007. Alteration of tomato fruit quality by root inoculation with plant growth-promoting rhizobacteria (PGPR): *Bacillus subtilis* BEB-13bs. *Scientia Horticulturae*. 113:103–106.

Mendelsohn, R., Nordhaus, W. and Shaw, D. 1994. Measuring the impact of global warming on agriculture. *American Economic Review*. 84:753–771.

Meng, Q., Jian, H., and Hao, J.J. 2016. Effect of *Bacillus velezensis* strain BACO3 in promoting plant growth. *Biological Control.* 98: 18-26.

Mohammed, H.A., Clark, J.A. and Ong, C.K. 1988. Genotypic difference in the temperature responses of tropical crops. I Germination characteristics of groundnut (*Arachis hypogea* L.) and pearl millet (*Pennisetum typhoides* S.&H.). *Journal of Experimental Botany*. 39: 1121-1128

Motsa, M.M., Slabert, M.M., van Averbeke, W. and Morey, L. 2015. Effect of light and temperature on seed germination of selected African leafy vegetables. *South African Journal of Botany*. 99: 29-35.

Muchuweti, M., Birkett, J.W., Chinyanga, E., Zvauya. R., Scrimshaw, M.D. and Lester, J.N. 2006. Heavy metal content of vegetables irrigated with mixture of wastewater and

sewage sludge in Zimbabwe: implications for human health. *Agriculture, Ecosystem* and *Environment.* 112: 41–48.

Müller, K., Linkies, A., Vreeburg, R.A.M., Fry, S.C., Krieger-Liszkay, A. and Leubner-Metzger, G. 2009. *In vivo* cell wall loosening by hydroxyl radicals during cress seed germination and elongation growth. *Plant Physiology*. 150: 1855-1865.

Naicker, K., Cukrowska, E. and McCarthy, T.S. 2003. Acid mine drainage arising from gold mining activity in Johannesburg. *South African Journal of Environmental Pollution.* 122: 29–40.

Naidoo, S. 2017. Acid Mine Drainage in South Africa, Springer Briefs in Environmental Science, DOI 10.1007/978-3-319-44435-2_2.

Ncube, B., Twomlow, S.J., Van Wijk, M.T., Dimes, J.P. and Giller, K.E. 2007. Productivity and residual benefits of grain legumes to sorghum under semi-arid conditions in south-western Zimbabwe. *Plant Soil*. 299: 1-15.

Nemutanzhela, M.V., Modise, D.M., Siyoko, K, J. and Kanu S.A. 2017. Assessment of growth, stomatal conductance and chlorophyll content of two potato genotypes under irrigation with Fly Ash-treated Acid Mine Drainage. *American Journal of Potato Research*. 94: 1-12.

Nevhulaudzi, T., Kambewa, C. and Kanu, S.A. 2014. Water quality around the Sheba gold mine: a case study of the Mlambomlwane River in Sheba, Mpumalanga Province, South Africa. *Scholarly Journal of Science Research and Essay*. <u>http://www.scholarly-journals.com/SJSRE</u>

Nevhulaudzi, T., Ntushelo K. and Kanu S.A. 2017. Interactive effect of *Bacillus subtilis* and elevated temperature on seed germination of three cowpea cultivars. State of the World's Plants Symposium, Royal Botanic Gardens, Kew, United Kingdom. Book of Abstract p24.

Niranjan, R.S., Deepak, S.A., Basavaraju, P., Shetty, H.S., Reddy, M.S. and Kloepper, J.W. 2003. Comparative performance of formulations of plant growth promoting *rhizobacteria* in growth promotion and suppression of downy mildew in pearl millet. *Crop Protection*. 22(4): 579-588.

Ntombela, Z. 2012. Growth and yield responses of cowpeas (*Vigna unguiculata* L.) to water stress and defoliation. Master's dissertation, University of KwaZulu-Natal, Durban.

Ofori, F. and Stern, W.R. 1987. Cereal-legume intercropping systems. *Advances in Agronomy*. 41: 41-90.

Okon, Y. and Labandera-Gonzalez, C.A. 1994. Agronomic applications of *Azospirillum*: an evaluation of 20 years worldwide field inoculation. *Soil Biology and Biochemistry*. 26: 1591–1601.

Olisa, B.S., Ajayi, S.A. and Akande, S.R. 2010. Physiological quality of seeds of promising African yam bean (*Sphenomstylis stenocarpa* (Hpechst. Ex. A. Rich) harms) and pigeon pea (*Cajanus cajan* L. Mill sp.) landraces. *Research Journal of Seed Sciences*. 3(2):93-101.

Oloya, T. and Tagwira, F. 1996. Land disposal of sewage sludge and effluent in Zimbabwe. Effect of applying sewage sludge and effluent on elemental accumulation and distribution in soil profile. *Zimbabwe Journal of Agriculture Research*. 34: 1-18.

Oyenuga, V.A. 1959. *Nigerian Feeding stuffs: Their compositions and nutritional value* 2nd Ed. University Press. p120.

Özkay, F., Kiran, S., Taş, I. and Kuşvuran, S. 2014. Effects of copper, zinc, lead and cadmium applied with irrigation water on some eggplant plant growth parameters and soil properties. *Turkish Journal of Agricultural and Natural Sciences*. 1: 377–383.

Padulosi, S. and Ng, N.Q. 1997. Origin, taxonomy, and morphology of Vigna unguiculata (L.) Walp. *Advances in Cowpea Research.* 1-12.

Pandey, N. and Sharma, C.P. 2002. Effect of heavy metals Co2+, Ni2+ and Cd2+ on growth and metabolism of cabbage. *Plant Science*. 163: 753-758.

Paredes-Páliz, K.I., Pajuelo, E., Doukkali, B., Caviedes, M.A., Rodríguez-Llorente, I.D. and Mateos-Naranjo, E. 2016. Bacterial inoculants for enhanced seed germination of Spartina densiflora: Implications for restoration of metal polluted areas. *Marine Pollution Bulletin*. 110(1): 396-400.

Platt, B.S. 1962. Tables of representative values of food commonly used in tropical countries. London: Medical Research Council special report, series 20.

Prajapati, M.R., Acharya, K.O. and Nawale, S. 2011. Impact of Climate Change on Food Security. *International Journal of Agriculture, Environment and Biotechnology*. 4(2): 125-127.

Prasad, P.V.V., Boote, K.J., Allen, L.H.J.R. and Thomas, J.M.G. 2002. Effects of elevated temperature and carbon dioxide on seed-set and yield of kidney bean (*Phaseolus vulgaris* L.). *Global Change Biology*. 8: 710–721.

Prathibha, K.S. and Siddalingeshwara, K.G. 2013. Effect of plant growth promoting *Bacillus subtilis* and *Pseudomonas fluorescens* rhizobacteria on seed quality of sorghum. *International Journal of Current Microbiology and Applied Sciences*. 2(3): 11-18.

Pukacka, S. and Ratajczak, E. 2005. Production and scavenging of reactive oxygen species in *Fagus sylvatica* seeds during storage at varied temperature and humidity. *Journal of Plant Physiology*. 162(8): 873-885.

Pule-Meulenberg, F., Belane, A.K., Krasova-Wade, T. and Dakora, F.D. 2010. Symbiotic functioning and bradyrhizobial biodiversity of cowpea (*Vigna unguiculata* L. Walp) in Africa. BMC Microbiolology, 10:89doi:10.1186/1471-2180-10-89

Quintana, J.M., Harrison, H.C., Neinhuis, J., Palta, J.P. and Grusak, M.A. 1996. Variation in calcium concentration among sixty S1 families and four cultivars of snap bean (*Phaseolus vulgaris* L.). *Journal of America Society of Horticultural Science*. 121: 789-793.

Rachie, K.O. and Silvestre, P. 1977. Grain legumes. In Leakey, C.L.A. and Wills, J.B. (Eds). *Food crops of the lowland Tropics*. London: Oxford University Press.

Rao, N.K. and Shahid, M. 2011. Potential of cowpea [*Vigna unguiculata* (L.) Walp.] and guar [*Cyamopsis tetragonoloba* (L.) Taub.] as alternative forage legumes for the United Arab Emirates. *Emirates Journal of Food and Agriculture*. 23 (2): 147-156.

Rennenberg, H., Loreto, F., Polle, A., Brilli, F., Fares, S., Beniwal, R. and Gessler, A. 2006. Physiological responses of forest trees to heat and drought. *Plant Biology*. 06(8): 556–571.

Rivers, W.G. and Weber, D.J. 1971. The Influence of Salinity and Temperature on Seed Germination in *Salicornia bigelovii*. *Physiology Plant*. 24: 73-75.

Román-Ponce, B., Reza-Vázquez, D.M., Gutiérrez-Paredes, S., DeHaro-Cruz, M.D., Maldonado-Hernández, J., Bahena-Osorio, J. Estrada-De Los Santos, P., En Tao-Wang, E.T. and Mvásquez-Murrieta, M.S. 2017. Plant growth-promoting traits in *Rhizobacteria* of heavy metal-resistant plants and their effects on *Brassica nigra* seed germination. *Pedosphere*. 27(3): 511-526.

Rusinamhodzi, L., Murwira, H.K. and Nyamangara, J. 2006. Cotton-cowpea intercropping and its N₂ fixation capacity improves yield of a subsequent maize crop under Zimbabwean rain-fed conditions. *Plant and Soil*. 287: 327-336.

Saeidnejad, A.H., Kafi, M. and Pessarakli, M. 2012. Evaluation of cardinal temperatures and germination responses of four ecotypes of *Bunium persicum* under different thermal conditions. *International Journal of Agriculture and Crop science*. 4(17): 1266-1271.

Saxena, J., Rana, G. and Pandey, M. 2013. Impact of addition of biochar along with *Bacillus* sp. on growth and yield of French beans. *Scientia Horticulturae*. 162: 351–356.

Schippers, B. 1992. *Prospects for management of natural suppressiveness to control soil borne pathogenes.* In: Biological Control of Plant Diseases, Progress and Challenges for the Future. NATO ASI Series Life Sciences. Tjamos. E. C., Papavizas. G. C., Cook, R. J. (eds.). *Plenum Press, New York.* 230: 21–34.

Schwartz, A.R., Ortiz, I., Maymon, M., Herbold, C.W., Fujishig, N.A., Vijanderan, J.A., Villella, W., Hanamoto, K., Diener, A., Sanders, E.R., DeMason, D.A. and Hirsch, A.M. 2013. *Bacillus simplex*—A little known PGPB with anti-fungal activity—alters pea legume root architecture and nodule morphology when coinoculated with *Rhizobium leguminosarum* bv. *Viciae. Agronomy*. 3: 595-620.

Seed world Cowpeas (SWC), 2017. *A Library for Food Security*. <u>http://seedworld.com/cowpeas-library-food-security</u>. (Accessed 03 May 2018).

Sershen, A., Perumal, B., Varghese, P., Govender, S. and Ramdhani, P.B. 2014. Effects of elevated temperatures on germination and subsequent seedling vigor in recalcitrant *Trichilia emetic*a seeds. *South African Journal of Botany*. 99: 153-162.

Sharma, S.N. and Prasad, R. 2003. Yield and P uptake by rice and wheat grown in a sequence as influenced by phosphate fertilization with diammonium phosphate and

Mussoorie rock phosphate with or without crop residues and phosphate solubilizing bacteria. *Journal of Agricultural Science*. 141: 359–369.

Sheng, X.F. and Xia, J.J. 2006.Improvement of rape (*Brassica napus*) plant growth and cadmium uptake by cadmium-resistant bacteria. *Chemosphere* 64:1036–1042.

Shishido, M., Massicotte, H.B. and Chanway, C.P. 1996. Effect of plant growth promoting *Bacillus* strains on pine and spruce seedling growth and mycorrhizal infection. *Annals of Botany*. 77:433-441.

Singh, B.B. and Ntare, B. 1985. Development of improved cowpea varieties in Africa. In: Singh, S.R. and Rachie, K.O. (Eds). *Cowpea research, production and utilization*. New York, Wiley. p. 106-115.

Singh, B.B., Ehlers, J.D., Sharma, B. and Freire Filho, F.R. 2002. Recent progress in cowpea breeding. In: Fatokun, C.A., Tarawali, S.A., Singh, B.B., Kormawa, P.M. and Tamò, M.(eds) Challenges and opportunities for enhancing sustainable cowpea production. Proceedings of the World Cowpea Conference III held at the International Institute of Tropical Agriculture (IITA), Ibadan, Nigeria, 4-8 September 2000. IITA, Ibadan, Nigeria. p. 22-40. ISBN 978-131-190-8.

Singh, R., Singh, P. and Sharma, R. 2014. Microorganism as a tool of bioremediation technology for cleaning environment: A review. *Proceedings of the International Academy of Ecology and Environmental Sciences*. 4(1): 1-6.

Sita, K., Sehgal, A., HanumanthaRao, B., Nair, R.M., Vara Prasad, P.V., Kumar, S., Gaur, P.M., Farooq, M., Siddique, K.H.M., Varshney ,R.K and Nayyar, H. 2017. Food legumes and rising temperatures: effects, adaptive functional mechanisms specific to reproductive growth stage and strategies to improve heat tolerance. *Frontier Plant Science*. 8(1658):1-30. DOI:10.3389/fpls.2017.01658

Smartt, J. 1985. Evolution of grain legumes. III. Pulses in the genus *Vigna*. *Experimental Agriculture*. 21(2): 87-100.

Smit, B. and Yunlong, C. 1996. Climate change and Agriculture in China. *Global Environmental Change*. *6*:205-214.

Soltangheisi, A., Rahman, Z.A. and Ishak, C.F. 2014. Interactive effect of zinc and manganese on growth, uptake response and chlorophyll content of sweet corn (Zea mays var saccharata). *Asian Journal of Plant Sciences*.13: 26–33.

Stone, P. and Nicolas, M., 1994. Wheat cultivars vary widely in their responses of grain yield and quality to short periods of post-anthesis heat stress. *Functional Plant Biology*. 21: 887–900.

Stratten, G.W.1987. Review. In: Environmental Toxicology. Hodson. E. (ed.). *Elsiever, Amsterdam*. p.85-94.

Streten-Joyce, C., Manning, J., Gibb, K.S., Neilan, B.A. and Parry, D.L. 2013. The chemical composition and bacteria communities in acid and metalliferous drainage from the wet–dry tropics are dependent on season. *Science of the Total Environment*. 443, 65279.

Sumner, M.E., Fey, M.V. and Noble, A.D. 1990. Nutrient status and toxicity problems in acid soils. In: Soil Acidity, Ulrich, B and Sumner, ME (Eds.) Berlin, *Springer-Verlag.* p.149-182.

Tarawali, S.A., Singh B.B., Gupta, S.C., Tabo, R., Harris, F., Nokoe, S., Ferna ndez-Rivera, S., Bationo, A., Manyong, V.M., Makinde, K. and Odion, E.C. 2002. Cowpea as a key factor for a new approach to integrated crop–livestock systems research in the dry savannas of West Africa. In: Fatokun, C.A., Tarawali, S., Singh, B.B., Kormawa, P.M. and Tamo, M. (eds) Challenges and opportunities forenhancing sustainable cowpea production. *International Institute of Tropical Agriculture, Ibadan, Nigeria*. p.233–251.

Tiller, K.G., Oliver, D.P., McLaughlin, M.J., Merry, R.H. and Naidu. R. 1997. Managing cadmium contamination of agricultural land. In: Iskandar, I.K., Adriano, D.C. (eds). Remediation of soils contaminated with metals. Northwood, England: *Science Reviews*. p.225-255.

United Nations Framework Convention on Climate Change (UNFCCC). 2013. Report of the Global Environment Facility to the Conference of the Parties. [Online]. Available at:http://www.unfccc.int. (Accessed 20 march 2018)

Vadapalli, V.R.K., Klink, M.J., Etchehebers, O., Petrik, L.F., Gitari, W., White, R.A., Key, D., and Iwuoha, E. 2008. Neutralization of acid drainage using fly ash, and

strength development of the resulting solid residues. *South African Journal of Science*. 104: 317-322.

Venter. A., Herselman, J.E., Van der Merwe, G.M.E., Steyn, C. and Beukes, D.J. 2001. Developing soil acidity maps for southAfrica. In: Plant Soil Interactions at Low pH: 5th *International Symposium Abstract.* p.21.

Verma, J.P., Jarnardan, Y., Tiwara, K.N., Lavakush and Singh, V. 2010. Impact of plant growth promoting rhizobacteria on crop production. *International Journal of Agricultural Research*. 5: 954-83

Wahid, A.; Gelani, S., Ashraf, M. and Foolad, M. 2007. Heat tolerance in plants: An overview. *Environmental and Experimental Botany*. 2007, 61, 199–223.

Wang, J., Fang, W., Yang, Z., Yuan, Y., Zhu, J. and Yu. H. 2007. Inter- and intraspecific variations of cadmium accumulation of 13 leafy vegetables species in a greenhouse experiment. *Journal of Agricultural and Food Chemistry*. 55: 9118–9123.

Wani, P.A. and Zainab, I.O. 2016. Effect of Chromium (Vi) Reducing *Bacillus* species PZ3 on the Growth of Pea Plants in Chromium Amended Soil. *Research Journal of Environmental Toxicology*. 10(3): 144-151.

Wani, P.A., Khan, M.S and Zaidi, A. 2008. Chromium reducing and plant growth promoting *Mesorhizobium* improves chickpea growth in chromium amended soil. *Biotechnology Letters*. 30: 159-163.

Wani, P.A., Khan, M.S. and Zaidi, A .2007. Synergistic effects of inoculation with nitrogen fixing and phosphate solubilizing *rhizobacteria* on the performance of field grown chickpea. *Journal of Plant Nutrition and Soil Science*. 170: 283-287.

Watson, R.T., Zinyoera, M.C. and Moss, R.H. 1997. The regional impacts of climate change: An assessment of vulnerability. A special report of the IPCC Working Group II. Cambridge: Cambridge University Press.

Weller, D.M. 1988. Biological control of soil borne plant pathogens in the rhizosphere with bacteria. *Annual Review Phytopathology*. 26:379-407.

White, J.W. and Monstes, R. C. 1993. The influence of temperature on seed germination in cultivars of common bean. *Journal of Experimental Botany*, 44,1975-1800.

Wu, Y., Feng, Y., Paré, P.W., Chen, Y., Xu, R. S., Wu, S., Wang, S., Zhao, Q., Li, H., Wang, Y., and Zhang, J. 2016. Beneficial soil microbe promotes seed germination, plant growth and photosynthesis in herbal crop *Codonopsis pilosula*. *Crop and Pasture Science*. 67, 91-98.

Zengin, F.K. and Munzuroglu, O. 2006. Toxic effects of cadmium (Cd⁺⁺) on metabolism of sunflower (*Helianthus annuus* L.) seedlings. Acta Agricultural Escandinavica Section B. *Plant Soil Science*. 56: 224–229.

Zhang, H., Murzello C, Sun. Y., Kim, M.S., Xie, X., Jeter, R.M., Zak, J.C., Dowd, S.E. and Paré, P.W. 2010. Choline and osmotic-stress tolerance induced in *Arabidopsis* by the soil microbe *Bacillus subtilis* (GB03). *Molecular Plant Microbe Interaction*. 23: 1097–1104.

Zhuang, X., Chen, J., Shim, H. and Bai, Z. 2007. New advances in plant growth promoting *rhizobacteria* for bioremediation. *Environmental International*. 33: 406-413.